General Exchange Tariff Section 31 3rd Revised Sheet 1 Replacing 2nd Revised Sheet 1 MILCS 2010 FUELOR Sarvios Communication

WIRELESS 911 SERVICE

(AT) 1. Description of Service

REC'D FEB 012000

- 1.1 Wireless 911 Service (W 911) is a service offering which routes wireless calls to specific Public Safety Answering Points (PSAPs) and provides a Mobile Directory Number (MDN) for callback information.
- 1.2 W 911 accesses Wireless Automatic Location Information (WALI) (the cell site receiving the 911 call).
- 1.3 The W 911 customer must be either:
 - A. A municipality or other state or local government unit, or an authorized agent of one or more municipalities or other state or local governmental units to whom authority has been lawfully delegated. The customer must be legally authorized to subscribe to the service and have public safety responsibility to respond to telephone calls from the public for emergency police and fire service within the telephone central office areas arranged for 911 Service calling, or
 - B. A licensed wireless carrier with approval of the appropriate 911 agency.

2. Explanation of Terms

<u>Wireless Administrative Site</u>: A location responsible for administration of wireless end user records associated with one or more wireless switches. This location has the computer hardware and software necessary to create and transmit wireless end user information to the W 911 database.

Wireless Automatic Location Identification (WALI): A system that identifies cell site location and associated information for cellular 911 calls.

Wireless Automatic Number Identification (WANI): A system that (1) identifies the billing account for a call in other applications, but for 911 systems, identifies the calling party; and (2) can also be used as a call back number.

<u>Wireless 911 Control Office</u>: The LEC central office providing tandem switching capabilities for 911 service calls from all end offices. It controls the switching of MDN and pseudo-Automatic Number Identification (pANI) information to a PSAP and also provides the Selective Routing service feature, Speed Calling features, Call Transfer services and certain maintenance functions for each PSAP.

Wireless 911 DataBase: A system of manual procedures and computer programs used to create, store, and update the data required to provide the Selective Routing (SR) and location text information.

MICOCUMPUCIIC COMMISSION Samao

b



CANCELLED

June 29, 2007 TO-2002-185 Missouri Public Service Commission

(AT)

Issued: FEBRUARY 1, 2000

Effective:

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

MAR 17 2000

RECEIVEDal Exchange Tariff Section 31 MAY 12 1992nd Revised Sheet 1 Replacing 1st Revised Sheet 1 and MAN PUBLIC SERVICE COMMAiginal Sheet 2 Original Sheet 3 Original Sheet 4 Original Sheet 5 Original Sheet 6 Original Sheet 7 1st Revised Sheet 8 1st Revised Sheet 9 1st Revised Sheet 10 1st Revised Sheet 11 1st Revised Sheet 12 1st Revised Sheet 13 **Original** Sheet 14 2nd Revised Sheet 15 Original Sheet 16 2nd Revised Sheet 17 Original Sheet 18 Original Sheet 19 Original Sheet 20 Original Sheet 21 2nd Revised Sheet 22 Original Sheet 23 Original Sheet 24 3rd Revised Sheet 25 1st Revised Sheet 26 2nd Revised Sheet 27

CANCELLED

MAR 1 7 2000 By 3 A R S I Public Service Commission MISSOURI

OBSOLETE SWITCHING SYSTEMS

The Obsolete Switching Systems offering has been withdrawn, in its entirety, and the contents completely removed.

FILED

JUN 21 1993

MO. PUBLIC SERVICE COMML

JUN 2 1 1993

Issued: MAY 1 7 1993

Effective: JUNIT SS

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.1 GENERAL REGULATIONS

NOV 3 0 1984

General Exchange Tariff

Replacing or teined Sheet

lst Revised Sheet 1

Section 31

- 31.1.1 This Tariff is applicable to obsolete Centrex C.U. Service served by No. 101 ESS Switching Vehicles that continue to be maintained by the Telephone Company.
- 31.1.2 Centrex C.U. The Telephone Company will determine and provide the necessary switching equipment and attendant positions required for exchange and other services furnished by the Telephone Company. The Telephone Company reserves the right to remove switching equipment furnished by the Telephone Company when, in its judgment, such equipment or facilities are no longer required or necessary to provide exchange or other services furnished by the Telephone Company. Provision of optional features is limited to the capabilities of the serving vehicle. See Paragraph 31.1.25 for additional charges applicable to Centrex C.U. Service.
- 31.1.3 Additional equipment or facilities required as a result of seasonal business requirements or for customer-owned facilities will be charged for on an individual-case basis.
- 31.1.4 The following terms as used in this Section apply to Centrex C.U. Service and shall mean:

Primary Centrex location - The customer's premises where the attendant equipment is located.

Secondary Centrex location - Customer's premises other than the location where the attendant equipment is located.

Centrex station line - a main station line of the Centrex System which has full in-out dialing privileges and access to or from the attendant.

Fully restricted station line - a main station line having intra-system dialing privileges only and having no access to or from the attendant. Fully restricted station lines may call any other station line of the system.

system.	CANCELLED	
	JUN 211993 #	
	BI Commission	FILED
	Effective DEC 311 BARRON, President-Missouri 2 hwestern Bell Telephone Comp	Division

St. Louis, Missouri

(RT)

OBSOLETE SWITCHING SYSTEMS

General Exchange Tariff Li E E E Section 31 Original Sheet 1 DEC 20 1833 MISSOURI Public Service Commission

- 31.1 GENERAL REGULATIONS
 - 31.1.1 This Tariff is applicable to obsolete Centrex C.U. Service served by No. 101 ESS Switching Vehicles that continue to be maintained by the Telephone Company and Communications Service Package Premises Systems Custom Calling Services.
 - 31.1.2 Centrex C.U. The Telephone Company will determine and provide the necessary switching equipment and attendant positions required for exchange and other services furnished by the Telephone Company. The Telephone Company reserves the right to remove switching equipment furnished by the Telephone Company when, in its judgment, such equipment or facilities are no longer required or necessary to provide exchange or other services furnished by the Telephone Company. Provision of optional features is limited to the capabilities of the serving vehicle. See Paragraph 31.1.25 for additional charges applicable to Centrex C.U. Service.
 - 31.1.3 Additional equipment or facilities required as a result of seasonal business requirements or for customer-owned facilities will be charged for on an individual-case basis.
 - 31.1.4 The following terms as used in this Section apply to Centrex C.U. Service and shall mean:

Primary Centrex location - The customer's premises where the attendant equipment is located.

Secondary Centrex location - Customer's premises other than the location where the attendant equipment is located.

Centrex station line - a main station line of the Centrex System which has full in-out dialing privileges and access to or from the attendant.

Fully restricted station line - a main station line having intra-system dialing privileges only and having no access to or from the attendant. Fully restricted station lines may call any other station line of the system.



FILED JAN - 1 1984 83-253 Public Service Commission

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

OBSOLETE SWITCHING SYSTEMS

1.1 GENERAL REGULATIONS-(Continued)

31.1.4 The following terms as used in this Section apply to Centrez Selvice ComMission Service and shall mean:-(Continued)

> Dormitory station line - a Centrex station line which terminates in a student's room in a dormitory of an accredited public or private educational institution that has Centrex Service for its administrative use.

Workable Station Lines - Centrex station lines available for customer use.

- 31.1.5 All operation at customer's premises is performed at the expense of the customer and must conform with the rules and regulations the Telephone Company may consider necessary to maintain a proper standard of service.
- 31.1.6 Other facilities, miscellaneous and supplemental equipment, requested by customers and not detrimental to this service or other services of the Telephone Company will be furnished in accordance with regulations and at the rates specified in the applicable section of this Tariff.
- 31.1.7 Sufficient floor space and the necessary power wiring, power outlets and commercial power for the operation of Centrex Service are furnished by the customer who assumes all responsibility for the safe condition of such floor space, wiring, outlets and power.

UN 81303 تان، الأن ثلاثيرة

General Exchange Tariff

Section-31-

STERE STEEL

DEU 20 203

MISSUURI



Issued: DEC 2 9 1983

Effective:

JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 Original Sheet 2 MICCOUNT PUBMIC Service Commission

RECTO FEB 01 2000

WIRELESS 911 SERVICE

2. Explanation of Terms - (Continued)

Mobile Directory Number (MDN): A 10-digit telephone number that identifies the calling party and can be used as a call back number.

<u>Pseudo-Automatic Number Identification (pANI)</u>: A number, consisting of the same number of digits as ANI, but is used in place of an ANI to convey a special meaning. The specific meaning assigned to the pANI is determined by agreements, as necessary, between the telephone system originating the call, intermediate telephone systems handling and routing the call, and the destination telephone system. The pANI identifies the location of the base station or cell site through which a mobile call originates.

Wireless Carrier: A private entity that provides telephone service to residential or business end users served by the provider's wireless switch.

Wireless Carrier Switch: A switch that provides wireless telephone service.

<u>Wireless End User</u>: An individual or organization authorized to use the telephone services provided by the wireless switch.

 Rules and Regulations – In addition to the following rules and regulation, the Rules and Regulations in the 911 Emergency Number Services Section of the Company's General Exchange Service shall also apply.

3.1 Application for Service

Request for this service: (1) can only be initiated by a 911 customer or a wireless carrier with approval of the 911 customer; (2) must be provided to Southwestern Bell Telephone Company (SWBT) in writing; and (3) must identify service locations and arrangements.

PSAP CPE must be capable of accepting 10+ digits when the W 911 tandem office is digital and 20+ digits when the W 911 tandem is analog. If a PSAP's CPE doesn't meet these requirements, Southwestern Bell Telephone Company will still deliver calls and routing for wireless calls on an 8- digit basis.

MICCOUN PUBLICA

FILED MAR 17 2000

CANCELLED

June 29, 2007 TO-2002-185 Missouri Public Service Commission

Issued: FEBRUARY 1, 2000

Effective:

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

MAR 17 2000

OBSOLETE SWITCHING SYSTEMS

No supplement to this tariff will be issued except for the purpose of canceling this tariff.



31.1 GENERAL REGULATIONS-(Continued)

- 31.1.8 One primary directory listing will be furnished without charge with each Centrex Service in accordance with the regulations set forth in Section 6 of this Tariff.
- 31.1.9 The primary location of a Centrex customer served by switching equipment located on customer premises must be in the same serving office area from which service is furnished.
- 31.1.10 A Centrex secondary location is furnished the same calling scope as the primary location. In addition, the switching equipment must be located in the same serving office area as the customer's secondary location.
- 31.1.11 Service at a secondary location may be provided with use of either switching equipment located on Telephone Company or customer's premises, irrespective of the method of service provided at the primary or other secondary locations within the Centrex System. [NOTE: Private Line Service Tariff charges are applicable to private lines (tie lines) between a Centrex C.U. premises vehicle and any other switching vehicle, including other Centrex C.U. or Centrex C.O. switching vehicles of the same Centrex System. In addition, Universal Tie Line Terminal Charges are applicable.]
- 31.1.12 Service furnished at secondary locations must be of the same grade and class of service as that furnished at the primary location of the same grade and
- 31.1.13 Where a Centrex C.O. secondary location is associated with a Centrex C.U. primary of the same Centrex System, rates applicable to the Centrex C.O. secondary location shall be equivalent to those for secondary location-- located in same serving office area as primary location.
- 31.1.14 A secondary location of a Centrex System will be furnished only if located in the same exchange as the primary location or zones of a metropolitan area, except as provided for by Optional and Extended Area Service arrangements as offered in this Tariff.
- 31.1.15 Centrex C.U. secondary locations are subject to Private Line Service Tariff charges and exchange trunking charges as-specified-in-Paragraph 31.1.25 of this Tariff.

F「「LED JAN - 1 1934 83 - 253 Public Service Commission

Issued: DEC 29 1983

Effective: JAN 0 1 1984

General Exchange Tariff Section 31 Original Sheet 3 MICCOVIN PUBLIC SOMMEDION

WIRELESS 911 SERVICE

REC'D FEB 012000

Rules and Regulations – In addition to the following rules and regulation, the Rules and Regulations in the 911
 Emergency Number Services Section of the Company's General Exchange Service shall also apply - (Continued)

P.S.C. Mo.-No. 35

- 3.2 Customer Obligations (1)(2)
 - A. The 911 customer is responsible for coordinating with SWBT and the wireless carrier so that the wireless switch provides the MDN in the CPN field of the SS7 signaling and pANI in the CDN (called number) field according to the technical specifications established by SWBT. The wireless switch number information must be approved by SWBT prior to implementation to ensure that the service will function properly. W 911 Service will not function properly if MDN and pANI are not in the proper fields, if duplicate telephone numbers exist at the wireless switch, or if any telephone numbers assigned by the wireless carrier are inconsistent with the SWBT's numbering plan.
 - B. The 911 customer and the wireless carrier are responsible for coordinating with SWBT so that the wireless carrier creates, maintains, and forwards to SWBT current MDN and pANI data according to the format and procedures specified by SWBT provided SWBT is the database provider.
 - C. The 911 customer and the wireless carrier are responsible for coordinating with SWBT so that the wireless carrier uses computer hardware and software for ongoing Wireless End User (WEU) record update programs and processes, that conform to the specifications outlined in the W 911 Technical Interface Standards.
 - D. W 911 Service WEU's information is confidential. The 911 customer agrees to use such information only for the purpose of responding to emergency calls.
 - E. The WEU forfeits the privacy afforded by nonlisted and nonpublished service to the extent that the telephone number, the address and name associated with the originating station location are furnished to the PSAP and to the LEC. The WEU (published and nonpublished) consents to the storage and retention of WEU name, location, telephone number, and address in the data base and also consents to access to this information by the PSAP for the sole purpose of responding to an emergency call.
 - F. The W 911 customer has the responsibility for reporting all errors, defects and malfunctions that they are aware of to SWBT in a timely manner provided SWBT is the service provider.
- (1) Customer obligations described are for situations where the E 911 customer is also the W 911 customer. In situations where a wireless carrier is the W 911 customer, the W 911 customer obligations described in this section are the direct responsibility of the wireless carrier.
- (2) SWBT will participate in all coordination efforts as appropriate.



CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission

Issued: FEBRUARY 1, 2000

Effective:

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

MAR 17 2000

General Exchange Tariff

DEC 20 1203

MISSOURI

Public Service Commission

Original Sheet 4

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.1 GENERAL REGULATIONS-(Continued)

- 31.1.16 The standard or optional features of a Centrex I C.U. and Centrex II C.U. System shall not be combined so as to provide any features of the higher grade system at the rate of the lower grade system.
- 31.1.17 The assignment of telephone numbers for Centrex Systems shall be in accordance with Section 17 of this Tariff.
- 31.1.18 When a customer requests special equipment or modification of standard equipment, it will be furnished whenever possible if not detrimental to any of the services or equipment of the Telephone Company. Such special equipment or modification of standard equipment will be furnished at additional charges.
- 31.1.19 Private line charges as specified in Section 2 of the Private Line Service Tariff apply to station lines extending outside the same building or beyond the same premises at both primary and secondary docations. Such station lines must be provided from the location from which they are measured.
 - Customer locations meeting the minimum station line requirements for secondary location rate treatment, but served by switching equipment serving another primary or secondary location, may be rated as secondary locations in lieu of mileage charges as specified above. Such locations must be in the same serving office area as the related primary or secondary and must be served by switching equipment located on Telephone Company premises.

Urban mileage charges for circuits extending beyond the base rate area apply at a primary location as specified in Section 12 of this Tariff. Such charges do not apply to secondary locations.

31.1.20 Mileage charges do not apply to Centrex dormitory station lines that are located in the same serving office area that serves the educational institution.

No minimum number of dormitory station lines is required to establish Centrex dormitory service, except that all occupied dormitory rooms in a dormitory served by Centrex dormitory service must be equipped with a Centrex dormitory station line. Charges for Centrex dormitory station lines do not apply to the minimum charge, per system.

31.1.21 When switching equipment is located on the customer's preprises, 5m2leage charges applicable to secondary locations are measured from the primary location.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 31 MICOCONTRIBUSTING4 Commission Borvies

RECT FEB 01 2000

WIRELESS 911 SERVICE

3. Rules and Regulations - In addition to the following rules and regulation, the Rules and Regulations in the 911 Emergency Number Services Section of the Company's General Exchange Service shall also apply - (Continued)

3.2 Customer Obligations - (Continued) (1)(2)

G. Cancellation of the service in whole or in part by the W 911 customer prior to establishment thereof, will require payment to SWBT of an amount equal to the cost of engineering, manufacturers' billings resulting from equipment orders, installation, assembly, labor, cost of removal and any other costs incurred by SWBT up to the time of cancellation resulting from the W 911 customer's order for service.

3.3 Liabilities

- SWBT's entire liability to any person for interruption or failure of W 911 Service shall be Α. limited by the terms set forth in this section and other sections of this tariff. SWBT or its officers or employees, may not be held liable for any claim, damage, or loss arising from the provision of W 911 Service unless it is proven that the act or omission proximately causing the claim, damage, or loss constitutes gross negligence, recklessness, or intentional misconduct on the part of SWBT.
- B. W 911 Service is provided solely for the benefit of the W 911 customer; the provision of such service shall not be interpreted, construed or regarded as being for the benefit of, or creating any LEC obligation toward, or any right of action on behalf of, any third person or other legal entity.
- C. To the extent allowed by law, the W 911 customer agrees to release, indemnify, defend and hold harmless SWBT from any and all loss, claims, demands, suits or other action, or any liability whatsoever, to any other party or person, for any personal injury to or death of any person or persons, or for any loss, damage or destruction of any property, whether owned by the W 911 customer or others.
- (1) Customer obligations described are for situations where the E 911 customer is also the W 911 customer. In situations where a wireless carrier is the W 911 customer, the W 911 customer obligations described in this section are the direct responsibility of the wireless carrier.
- MIMONT PUBLIC (2) SWBT will participate in all coordination efforts as appropriate.

FILED MAR 17 2000

Issued: FEBRUARY 1, 2000

Effective: Manual Manual Manual

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

MAR 17 2000

CANCELLED June 29, 2007 TO-2002-185 **Missouri Public** Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.



OBSOLETE SWITCHING SYSTEMS

31.1 GENERAL REGULATIONS-(Continued)

- 31.1.22 Centrex main lines terminating in attendant positions or dial equipment of another PBX, Centrex or similar system constitute station-to-trunk tie lines. Universal Tie Line Terminal rates are applicable to such terminations. The Centrex station line rate is not applicable to station-to-trunk tie lines.
- 31.1.23 Tie Line Terminal rates specified in Paragraph 31.6 of this Tariff apply for each tie line terminating in a Centrex System.
- 31.1.24 Network Access Terminal rates shall apply for each SCAN- and CCSA-type access line terminating in a Centrex System.
- 31.1.25 Standard Centrex C.U. features specified in Paragraph 31.3.1, following, require services provided in other sections of this Tariff and other tariffs, including Local Exchange Tariff and the Private Line Service Tariff. These services are provided at standard Tariff rates and include:
 - Direct Inward Dialing Service to Customer Premises Located Switching Systems (Private Branch Exchange Service - General Exchange Tariff).
 - Automatic Identified Outward Dialing Service From Customer -Premises Located Switching Systems (Private Branch Exchange Service - General Exchange Tariff).
 - Exchange Trunking (Local Exchange Tariff).
 - Tie lines between a Centrex C.U. premises vehicle and any other switching vehicle (Private Line Service Tariff).
 - Touch-Tone Calling Service (General Exchange).

CANCELLED JUN 8:11583 EV 2 ~ R.J. # FILED האנטווס נפריוטב ככוווואני JAN - 1 1984 83-253 Public Service Constantion

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

General Exchange Tariff Section 31 Original Sheet 5 MICCOVIN MUSING BOMAGE COMMICIDIOR

WIRELESS 911 SERVICE

RECD FEB 0 1 2000 Rules and Regulations - In addition to the following rules and regulation, the Rules and Regulations in 3. the 911 Emergency Number Services Section of the Company's General Exchange Service shall also apply - (Continued)

3.3 Liabilities (Continued)

- D. To the extent allowed by law, the W 911 customer also agrees to release, indemnify and hold harmless SWBT for any infringement or invasion of the right of privacy of any person or persons, caused or claimed to have been caused, directly or indirectly, by the installation, operation, failure to operate, maintenance, removal, presence, condition, occasion or use of 911 service features and the equipment associated therewith, or by any services furnished by SWBT in connection therewith, including, but not limited to, the identification of the telephone number, address or name associated with the telephone used by the party or parties accessing W 911 service hereunder, and which arise out of the negligence of SWBT or the negligence or other wrongful act of the W 911 customer, its user, agencies or municipalities or the employees or agents of any one of them.
- W 911 Service will be designed by SWBT to provide at least the same level of service reliability E. and quality as local exchange telephone service in the exchanges where 911 systems are equipped with the features required to provide W 911 Service.
- F. Adjustments for service interruptions experienced by the W 911 customer are governed by and limited in the Rules and Regulations Applying to All Customers' Contracts Section of the General Exchange Tariff.
- G. To the extent allowed by law, the W 911 Customer and the Wireless Carrier agree to indemnify, defend, and hold harmless SWBT from any and all claims, costs, demands, liabilities, losses, actions, lawsuits, and expenses arising from W 911 Customer and/or Wireless Customer providing SWBT with inaccurate, out of date, or improperly formatted MDN or pANI data.
- H. To the extent allowed by law, the W 911 Customer and the Wireless Carrier agree to indemnify, defend, and hold harmless SWBT from any and all claims, costs, demands, liabilities, losses, actions, lawsuits, and expenses arising from any and all equipment failure or defects or errors in transmission on the part of the W 911 customer or the wireless carrier.

BOMEO COmmunication

FILED MAR 17 2000

CANCELLED June 29, 2007 TO-2002-185 **Missouri Public** Service Commission

FEBRUARY 1, 2000 Issued:

Effective:

By JAN NEWTON, President-Missouri MAR 17 2000 Southwestern Bell Telephone Company St. Louis, Missouri

General Exchanges Tariff Section 31 Original Sheet DEC 20 1983 I.I.SSOURI Public Service Commission

OBSOLETE SWITCHING SYSTEMS

31.1 GENERAL REGULATIONS-(Continued)

31.1.26 Centrex C.U. - The Exchange Access and Intercommunication portion of main station line charges are not offered separately. The rates and charges specified herein for main station lines provide for a main station line component. The main station line consists of all facilities, including intercommunication, from the Centrex system dial switching equipment to the demarcation point for connection to the premises wire.

> The rates and charges specified herein for fully restricted main station lines provide for a fully restricted main station line component. The fully restricted main station line consists of all facilities, including intercommunication, from the Centrex system dial switching equipment to the demarcation point for connection to the premises wire.

- 31.1.27 Centrex I and Centrex II Service served by switching equipment located on the customer's premises is obsolete to existing installations at existing locations.
 - Except that, growth is permitted only to the capacity of the existing vehicle.
 - Except that, with a transfer of contract, growth will be permitted only to the capacity of the existing vehicle.

Where the primary location is served by switching equipment located on customer premises, system growth in the form of additional secondary locations (C.O. or C.U.) is not permitted.

This obsolescence is applicable to each primary and each secondary location.

31.2 CENTREX C.U. - WHERE REFERENCE IS MADE TO "NRX" IN THIS TARIFF, IT DENOTES THE STATION LINE RATE FOR THE PREMISES VEHICLE

CANCELLED JUN 211803 FILED Public Service Committee JAN - 1 1984 83 - 253 ubli Service Contraine

Issued: DEC 29 1983

Effective: JAN 0 1 1984

General Exchange Tariff Section 31 Original Sheet 6

WIRELESS 911 SERVICE

P.S.C. Mo.-No. 35

Some Communication

RFCD FEB 012000

4. Wireless 911 Service Rates and Charges

4.1 W 911 Service

No supplement to this

except for the purpose of canceling this tariff.

tariff will be issued

		Nonrecurring		
		Monthly	Service	
		Rate	Charge	USOC
A.	W 911 Service with			
	Selective Routing			
	Charge per $pANI(1)(2)(3)$	\$ 24.00	\$30.00	W9A1X
B.	Selective Routing			
	Charge per pANI(1)(2)	\$ 12.00	\$10.00	W9S1X
	0. F. T ()(-)			
C.	Automatic Location			
	Identification			
	Charge per pANI (3)	\$ 12.00	\$20.00	W9L1X
	5 1 1 1 1 2 2	-		

4.2 Database Management System

CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission The rates and charges in this section will equal the Database Management System monthly rate and nonrecurring charge in Paragraph 28.7.4.B (Database Management System) in Section 28 (Universal Emergency Number Services (911)) of the General Exchange Tariff. These rates and charges relate to the administration and storage of W 911 Service pANI records used for routing and associated location text. The monthly rate will be per 10 records per W 911 customer.

A nonrecurring charge equal to the nonrecurring charge to establish an administrative site in Paragraph 28.7.4.B (Database Management System) in Section 28 (Universal Emergency Number Service (911)) of the General Exchange Tariff applies at the time an administrative site is first established. This charge applies when procedures are established enabling the administrative site to create and update records.

- (1) There will be a trunk needed between the Mobile Switching Center (MSC) and the 911 tandem. The wireless carrier, in cooperation with the appropriate 911 agency, will purchase that element out of the Wireless Carrier Interconnection Service tariff or its existing interconnection agreement.
- (2) When the 911 agency involved also uses SWBT as its 911 database and network provider, the calls and data will be sent over the existing tandem to PSAP trunks and ALI circuits. If the 911 agency uses a different network provider or database provider, the tandem to PSAP trunks and/or ALI circuits must be provisioned separately.
- (3) When SWBT is the Wireless 911 Service provider, both W 911 Service with Selective Routing and Automatic Location Identification rates and charges, paragraphs 4.1.A and 4.1.C will be required. The Selective Routing rates and charges in paragraph 4.1.B only apply if an alternative Wireless 911 Service provider is utilized.

		Miccount Public Sorrico Commiccio
Issued:	FEBRUARY 1, 2000	Effective: FILED MAR 1 7 2000
		By JAN NEWTON, President-Missouri MAR 17 2000
		Southwestern Bell Telephone Company
		St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.3 CENTREX C.U. SERVICE - STANDARD FEATURES

31.3.1 Centrex Service served by switching equipment located on customer premises (C.U.).

> Centrex I includes the following standard features(1)(2)(3)(7): attendant positions, direct outward calling, direct inward calling, station-to-station calling, station hunting, call transfer-attendant, restriction from outgoing calls, station restriction(4), reserve power, busy verification, night service, identified outward calling and call interception.

Centrex II(5)(6)(7) - In addition to all standard features provided with Centrex I C.U. Service, the following additional standard features are provided: call transfer-individual, consultation hold, add-on, trunk answer from any station and Touch-Tone Calling Service.

CANCELLED

General Exchange Vie

Section 31

Original Sheet 7 DEC 20 1933

MISSOURI **Public Service Commission**

JUN 811223 #1 Phoble Cervice Committeeuri

- (1) Station hunting is limited to not more than ten station lines with each hunting arrangement. Requests for service above ten consecutive station lines may be provided on a Special Service Arrangement Request basis.
- (2) The reserve power feature provides an alternate independent source of power to maintain telephone service (excluding certain attendant supervisory functions and key telephone arrangements) during a commercial power failure. The Telephone Company agrees to make every reasonable effort to assure continued operation until such time as commercial power is restored.
- (3) Busy verification will be provided subject to the operating limitations of the equipment or facilities selected by the Telephone-Gompany and providedno unusual expense is involved.
- (4) Does not include toll restriction.
- (5) Call transfer-individual, consultation hold and add-on will function only incoming call from the telecommunications network. JAN I 1984 on an incoming call from the telecommunications network.
- (6) Add-on provided for a maximum of one additional station lin8 2 in 2 a dd at one time to an existing incoming call from the telecommunications metwork (7) See Paragraph 31.1.25 for additional charges applicable to Centrex C.U. Service.

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

General Exchange Tariff Section 31 Original Sheet 7

WIRELESS 911 SERVICE

SUM F

5. TERM PRICING PLAN

5.1 General

REC'9 FEB 012000

- A. Wireless 911 Service Term Pricing Plan (W911-TPP) provides the customer with rate stabilization and discounted tariff rates. The W911-TPP provides for either a one or three year service period (Initial Service Period) for rate stabilization.
- B. W911-TPP monthly rates will be exempt from Telephone Company initiated rate increases throughout the selected service period. Should the Telephone Company increase its rates during the W911-TPP period, the customer would continue to pay the rates in effect at the time the customer elected to establish service under W911-TPP.
- C. Decreases in W911-TPP monthly recurring tariff rates will be passed on to customers who participate in a W911-TPP.
- 5.2 The monthly charge per pANI is available under W911-TPP.
- 5.3 Terms and Conditions
- A. The customer must specify the length of the Initial Service Period at the time the service is ordered.

Changes to Pricing Plans

- B. At any time, the customer may request existing W 911 provided on a monthly rate basis to be converted to a W911-TPP.
- C. Prior to the expiration of the Initial Service Period or Extended Service Period described in 5.5 (Renewal), the customer may convert existing W911-TPP services to a new W911-TPP initial Service Period without incurring termination charges provided the new Initial Service Period is equal to or greater than the original Initial Service Period.
 - Example: A customer with an existing <u>one year</u> W911-TPP could convert to a new <u>one or three year</u> W911-TPP at any time without incurring termination charges.
 - Example: A customer with an existing three year W911-TPP requests to convert to a <u>one year</u> W911-TPP. This request would be treated as a discontinuance of the existing three year W911-TPP and termination charges would apply.
- D. If a customer requests existing W911-TPP service to be converted to a monthly rate basis at any time prior to the expiration of the Initial Service Period or Extended Service Period, the request will be treated as a discontinuance of service and termination charges will apply MICOSUM 1905 Mice

Issued: FEBRUARY 1, 2000

CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission Effective

r 17 2000

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

5.5

General Exchange Tariff Section 31 Original Sheet 8

WIRELESS 911 SERVICE

5. TERM PRICING PLAN - (Continued)

ച്ചാത്ര

REC'D FEB 012000

Renewal

- A. The customer may elect to renew the W911-TPP Initial Service Period for one additional 12 month service period (Extended Service Period) at the rates under the original W911-TPP Initial Service Period.
- B. The customer must provide the Telephone Company with a written notice of intent to renew an existing W911-TPP Initial Period no later than 90 days prior to the expiration of the Initial Service Period.
- C. If the customer elects not to renew the W911-TPP or does not notify the Telephone Company of its intent to renew, the customer's service will automatically be billed under the tariffed monthly rates in effect at the time the W911-TPP Initial Service Period expires.
- 5.6 Moves to New Location
 - A. A customer with an existing W911-TPP service may move the existing service to a new location without incurring termination charges provided all of the following conditions are met:
 - the new service is provided solely by the Telephone Company;
 - the new location is within the same LATA;
 - the customer's request for disconnection of the existing and the request for new service are received at the same time;
 - the due date of the new connect order must be within 30 days of the due date of the disconnect order;

FI FO MAR 1 7 2000

Issued: FEBRUARY 1, 2000

Effective:

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission

General Exchange Tariff Section 31 1st Revised Sheet 8 Replacing Original Sheet 8

OBSOLETE SWITCHING SYSTEMS

RECEIVED

31.4 RATES - CENTREX C.U. SERVICE(1)

SEP 2 5 1989

31.4.1 Centrex I Premises Vehicle Service

A. Primary Location

MISSOURI Public Service Commission

Monthly Rate(2)			
First 200	Next 300	0ver 500	
Station Lines	Station Lines	Station Lines	
(RXR) (RX2)	(RXR) (RX2)	(RXR) (RX2)	
(RUV) (RX5)(5)	(RUV) (RX5)(5)	(RUV) (RX5)(5)	
NRXXA	NRXXB	NRXXC	
\$ 4.15	\$3:35	\$1.95	

	Monthly <u>Rate</u>
Trunks, with rotary dial operation	(3)
Minimum Charge, per system	(4)

CANCELLEE JUN 817833 BY 2 R.S Fublic Certice C.

- (1) Refer to Paragraph 31.1.27 for obsolescence regulation.
- (2) Refer to Section 13 of this Tariff for applicable charges for DID and AIOD Services.
- (3) Refer to the Local Exchange Tariff for applicable PBX trunk rates.
- (4) The minimum charge for Centrex station lines served by switching equipment located on customer's premises, per Centrex System, shall be the rate applicable to 200 main Centrex station lines at the primary location and 50 main Centrex station lines at each secondary location.

Effective:

(CR) (5) A Service and Equipment Charge of \$28.50 applies for each main station line provided.

FILED

Issued: SEP 25 1989

06191 - 1989OCT 01 1989 By R. D. BARRON, President-Missouri Division Division Southwestern Bell Telephone Company St. Louis, Missouri

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)

31.4.1 Centrex I Premises Vehicle Service

A. Primary Location



First 200	Next 300	Over 500
Station Lines	Station Lines	Station Lines
(RXR) (RX2)	(RXR) (RX2)	(RXR) (RX2)
(RUV) (RX5)(5)	<u>(RUV) (RX5)(5)</u>	(RUV) (RX5)(5)
NRXXA	NRXXB	NRXXC
\$ 4.15	\$3.35	\$1.95
		Monthly
		Rate
inks, with rotary	dial operation	(3)
nimum Charge, per	system	(4) •

Monthly Rate(2)

CANCELLED

OCT 1 1989 Public Service Commission MISSOURI

(1) Refer to Paragraph 31.1.27 for obsolescence regulation.

- (2) Refer to Section 13 of this Tariff for applicable charges_for DID_and AIOD____ Services.
- (3) Refer to the Local Exchange Tariff for applicable PBX trunk rates $||\zeta||$ (4) The minimum charge for Centrex station lines served by switching equipment
- located on customer's premises, per Centrex System, shall be the Vräte applicable to 200 main Centrex station lines at the primary location god 205ng in Centrex station lines at each secondary location.
- (5) A Service and Equipment Charge of \$4.75 applies for each main station Gimession provided.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

General Exchange Tariff Section 31 Original Sheet 9

WIRELESS 911 SERVICE

P.S.C. Mo.-No. 35

RECO FEB 01 2000

- 5. TERM PRICING PLAN (Continued)
- 5.6 Moves to New Location (Continued)
- B. In the event an order to move service provided under W911-TPP does not meet one or more of the conditions above, it will be treated as a discontinuance of the existing service and the establishment of a new service. All termination charges will apply.
- C. Except as noted above, the monthly rates for the new service will be those in effect at the time the service is changed. All nonrecurring charges associated with the establishment of the new service will apply.

6. <u>TERM PRICING PLAN – RATE APPLICATIONS</u>

- 6.1 Nonrecurring Charges
- A. The nonrecurring service charges as described in 7.1 will apply for new services ordered under W911-TPP.
- B. If the customer chooses to convert an existing service provided on a W911-TPP to a monthly rate basis no nonrecurring charges will apply. However, W911-TPP termination charges may apply.
- 6.2 Termination Charges
- A. Customers requesting to discontinue services provided under a W911-TPP, prior to the expiration of the Initial Service Period or Extended Service Period will incur termination charges. Payment of the termination charge does not release the customer from other previous amounts owed to SWBT.

SONAGO

7 2000

Issued: FEBRUARY 1, 2000

CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission Effective:

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri MAR 17 2000

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 1st Revised Sheet 9 Replacing Original Sheet 9

Monthly

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

RECEIVED

31.4.1 Centrex I Premises Vehicle Service-(Continued)

B. Secondary Location

MISSOURI Public Service Commission

SEP 25 1989

First 200	Next 300	Over 500
Station Lines	Station Lines	Station Lines
(RXR) (RX2)	(RXR) (RX2)	(RXR) (RX2)
(RUV) (RX5)(6)	(RUV) (RX5)(6)	(RUV) (RX5)(6)
NRX++(5)	NRX++(5)	NRX++(5)
\$4.15	\$3.35	\$1.95

	Rate	CANCEL
Trunks, with rotary dial operation	(3)	UN 211
Minimum charge, per system	(4)	· 2nd R

- (1) Refer to Paragraph 31.1.27 for obsolescence regulation.
- (2) Refer to Section 13 of this Tariff for applicable charges for DID and AIOD Services.
- (3) Refer to the Local Exchange Tariff for applicable PBX trunk rates.
- (4) The minimum charge for Centrex station lines served by switching equipment located on customer's premises, excluding extension station lines and any other chargeable items of equipment or service, per Centrex System, shall be the rate applicable to 200 main Centrex station lines at the primary location and 50 main Centrex station lines at each secondary location.
- (5) The following NRX codes are applicable to Centrex C.U. secondary locations:

						1st 200	Next 300	0ver 500			
(CR)	(6)	Contig Noncon	itiguous vice and	rving Off Serving	Office	NRXXD NRXXG NRXXK of \$28.50 app]	NRXXE NRXXH NRXXL lies for	NRXXF NRXXJ NRXXM each main	station [FIL]		
	Issu	ued:	j E₽	25 1900		Effective:	OCT	011989.	0CT 1 89	1989 - 1 4	
						President-Miss Bell Telephon	souri Div ne Compar	vision ^{Publi} 1y	c Service	- 1 4 Commissic	1

St. Louis, Missouri

General Exchange Tariff

RECEIVED 9

DEC 2 0 1083

MISSOURI

Public Service Commission

-Section 31

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

31.4.1 Centrex I Premises Vehicle Service-(Continued)

B. Secondary Location

Monthly Rate(2)	
First 200 Next 300 Over 5 Station Lines Station Lines Station (RXR) (RX2) (RXR) (RX2) (RXR) (RUV) (RX5)(6) (RUV) (RX5)(6) (RUV) (1) NRX++(5) NRX++(5) NRX++(5)	Lines (RX2) RX5)(6)
\$4.15 \$3.35 \$1.9	95
Mont. Rat	•
Trunks, with rotary dial operation (3)
Minimum charge, per system : (4	CANCELLED
	OCT 1 1989 BY ARS#9 Ilc Service Commission MISSOURI
(1) Refer to Paragraph 31.1.27 for obsolescence regulation.	
(2) Refer to Section 13 of this Tariff for applicable charg Services.	es for DID and AIOD
 (3) Refer to the Local Exchange Tariff for applicable PBX t (4) The minimum charge for Centrex station lines served by located on customer's premises, excluding extension sta other chargeable items of equipment or service, per Cen be the rate applicable to 200 main Centrex station line location and 50 main Centrex station lines at each seco 	switching equipment tion lines and any trex System, shall s at the primary

(5) The following NRX codes are applicable to Centrex C.U. secondary Flocations:

		1st	Next	Over	าย และเลย
		200	<u>300</u>	<u>500</u>	JAN - 1 1984
·S	ame Serving Office	NRXXD	NRXXE	NRXXF	83-253
C	Contiguous Serving Office	NRXXG	NRXXH	NRXXJ	ic Service Comminsion
N	Ioncontiguous Serving Office	NRXXK	NRXXL	NRXXM	
(6) A	Service and Equipment Charge	of \$4.75 appli	es for	each main	station line

(provided.

DEC 2 9 1983 Issued:

JAN 0 1 1984 Effective:

General Exchange Tariff Section 31 Original Sheet 10

WIRELESS 911 SERVICE

Miccount Pulting Sowies Commission

6. <u>TERM PRICING PLAN - RATE APPLICATIONS - (Continued)</u>

- 6.2 Termination Charges (Continued)
 - B. The termination charge for the Initial Service Period shall be the lesser of:
 - The difference between the Initial Service Period rates and charges for the completed months of the initial service term at the time of termination and the rates and charges for the next lower service term(1) actually completed plus interest charges based on approved costs of money in effect at the time of termination; or
 - The present value of monthly payments remaining on the service term.
 - Example: If the customer completes 26 months of a 36 month (3 year) service term, the first calculation of a termination charge will equal the difference between 36 months of rates and charges at the 3 year service term and 26 months of rates and charges at the 1 year service term (which is the next lower service term actually completed), plus interest at the approved cost of money rate in effect at the time of termination. The second calculation will be the sum of the 10 remaining monthly payments of the 3 year service term. The termination charge is the lesser of the two calculations.
 - C. The termination charges for the Extended Service Period shall be the lesser of:
 - The difference between the extended service period rates and the month-to-month rates for the months actually completed plus interest charges based on approved cost of money in effect at the time of termination; or
 - The present value of monthly payments remaining on the extended service term.
 - D. If special construction was applied to the service being terminated, any termination charges associated with the special construction will also apply.
- (1) If the service is terminated within the first 12 months the calculation is based on month-tomonth rates and charges.

Issued: FEBRUARY 1, 2000

Effective: Book and Market Market

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

CANCELLED June 29, 2007 TO-2002-185 Missouri Public

Service Commission

P.S.C. Mo.-No. 35

RECD FEB 012000

SOMOG

MAR 17 2000

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 1st Revised Sheet 10 Replacing Original Sheet 10

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

31.4.1 Centrex I Premises Vehicle Service-(Continued)

C. Services for Educational Institutions

MISSOURI Public Service Commission

SEP 2 5 1989

Centrex I Service for dormitories administered by the educational institution may be provided at the following rates when associated with a Centrex System as provided by this Tariff.

Main Station lines

	Monthly	Rate(2)
(PSA)	(RV3)(3)(5)	(PS7) (RTX)(4)(5)
Ś	9,40	\$8.45

CANCELLED JUN 211833 EY 2 ~ (R.S. #1 Public Carvis C.

OCT 1 1989

(1) Refer to Paragraph 31.1.27 for obsolescence regulation.

(2) The customer shall elect the method in which billing is to be administered, and all dormitory station lines shall be billed accordingly.

(3) The Telephone Company will bill and collect toll charges.

(4) The Institution will bill and collect all charges.

(CR) (5) A Service and Equipment Charge of \$28.50 applies for each main station line provided.

Issued:SEP 25 1900

By R. D. BARRON, President-Missouri Division Division Commission Southwestern Bell Telephone Company St. Louis. Missouri

Effective:

OCT 01 1989

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

31.4.1 Centrex I Premises Vehicle Service-(Continued)

C. Services for Educational Institutions

Centrex I Service for dormitories administered by the educational institution may be provided at the following rates when associated with a Centrex System as provided by this Tariff.

Main Station lines

•		Monthly	Rate(2)
	(PSA)	(RV3)(3)(5)	Rate(2) (PS7) (RTX)(4)(5)
	\$	9.40	\$8.45

CANCELLED OCT 1. 1989 BY PIPSHID Public Service Commission MISSOURI

- Refer to Paragraph 31.1.27 for obsolescence regulation ublic Service Commission
 The customer shall elect the method in which billing is to be administered. and all dormitory station lines shall be billed accordingly.
- (3) The Telephone Company will bill and collect toll charges.
- (4) The Institution will bill and collect all charges.
- (5) A Service and Equipment Charge of \$4.75 applies for each main station line provided.

DEC 29 1983 Issued:

JAN 0 1 1984 Effective:

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

	DEC 29 1083
F	MISSOURI Public Service Commission

FILED

JAN - 1 1984 83-253

General Exchange Tariff



General Exchange Tariff Section 31 Original Sheet 11

WIRELESS 911 SERVICE

7. TERM PRICING PLAN – Rates and Charges

7.1 W 911 Service

RECD FEB 0 1 2000

			Nonrecurring	
		Monthly	Service	
		Rate	Charge	USOC
A.	W 911 Service with Selective Routing Charge per pANI (1)(2)(3)			<u>,</u>
	One Year Contract	\$ 21.00	\$10.00	W9A1X
	Three Year Contract	\$ 19.00	N/A	W9A1X
B.	Selective Routing Charge per pANI (1)(2)			
	One Year Contract	\$ 8.50	\$ 5.00	W9S1X
	Three Year Contract	\$ 5.00	N/A	W9S1X
C.	Wireless Automatic Location Identification Charge per pANI (3)			
	One Year Contract	\$ 10.00	\$10.00	W9L1X
	Three Year Contract	\$ 6.00	N/A	W9L1X

- There will be a trunk needed between the MSC and the 911 tandem. The wireless carrier, in (1)cooperation with the appropriate 911 agency, will purchase that element out of the Wireless Carrier Interconnection Service tariff or its existing interconnection agreement.
- (2) When the 911 agency involved also uses SWBT as its 911 database and network provider, the calls and data will be sent over the existing tandem to PSAP trunks and ALI circuits. If the 911 agency uses a different network provider or database provider, the tandem to PSAP trunks and/or ALI circuits must be provisioned separately.
- When SWBT is the Wireless 911 Service provider, both W 911 Service with Selective Routing (3) and Automatic Location Identification rates and charges, paragraphs 7.1.A and 7.1.C will be required. The Selective Routing rates and charges in paragraph 7.1.B onty apply of an alternative BOWIES COMMICOION Wireless 911 Service provider is utilized.

<u>ÁŘ 172000</u>

Issued: FEBRUARY 1, 2000

Effective:

By JAN NEWTON, President-Missouri MAR 17 2000 Southwestern Bell Telephone Company St. Louis, Missouri

CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 31 1st Revised Sheet 11 Replacing Original Sheet 11

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

RECEIVED SEP 2 5 1989

31.4.2 Centrex II Premises Vehicle Service

A. Primary Location

MISSOURI **Public Service Commission**

	Monthly Rate(2)	
First 200 Station Lines (RXR) (RX2) (RUV) (RX5)(5) <u>NRXXA</u>	Next 300 Station Lines (RXR) (RX2) (RUV) (RX5)(5) <u>NRXYB</u>	Over 500 Station Lines (RXR) (RX2) (RUV) (RX5)(5) NRXYC
\$4.80	\$4.00	\$2.80
		Monthly _Rate
Trunks, with rotary d	ial operation	(3)
Touch-Tone operation charge for trunks) .	-	
Minimum charge. per s	ystem	(4) CANCELLED
		UNN 211993 $E = \frac{2}{2} \frac{2}{2} \frac{2}{2} \frac{2}{2} \frac{1}{2} $

(1) Refer to Paragraph 31.1.27 for obsolescence regulation.

- (2) Refer to Section 13 of this Tariff for applicable charges for DID and AIOD Services.
- (3) Refer to the Local Exchange Tariff for applicable PBX Trunk rates.
- (4) The minimum charge for Centrex station lines served by switching equipment located on customer's premises, excluding extension station lines and any other chargeable items of equipment or service, per Centrex System, shall be the rate applicable to 200 main Centrex station lines at the primary location and 50 main Centrex station lines at each secondary location.
- (CR) (5) A Service and Equipment Charge of \$28.50 applies for each main station line provided.
 - (6) Refer to Section 24 of this Tariff for applicable charges.

FILED

89-14

Issued:

OCT 01 1990 ublic Service Commission SEP 25 1990 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis. Missouri

Effective:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

Monthly Rate(2)

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

31.4.2 Centrex II Premises Vehicle Service

A. Primary Location

General Exchange Tariff Section_31 REGENEED DEC 29 1633 []]SSOURI Public Service Commission

FILED

JAN - 1 1934

Next 300 First 200 Over 500 Station Lines Station Lines Station Lines (RXR) (RX2) (RXR) (RX2)(RXR) (RX2)(RUV) (RX5)(5)(RUV) (RX5)(5) (RUV) (RX5)(5)NRXYC NRXXA NRXYB \$4.80 \$2.80 \$4.00 Monthly Rate Trunks, with rotary dial operation . . . (3) Touch-Tone operation (in addition to charge for trunks) (6) CANCELLED Minimum charge. per system . (4)OCT 1 1989

(1) Refer to Paragraph 31.1.27 for obsolescence regulation.

- DID and 253 (2) Refer to Section 13 of this Tariff for applicable charges for DLD and Commission AIOD Services.
- (3) Refer to the Local Exchange Tariff for applicable PBX Trunk rates.
- (4) The minimum charge for Centrex station lines served by switching equipment located on customer's premises, excluding extension station lines and any other chargeable items of equipment or service, per Centrex System, shall be the rate applicable to 200 main Centrex station lines at the primary location and 50 main Centrex station lines at each secondary location.

Public Service Commission

MISSOURI

- (5) A Service and Equipment Charge of \$4.75 applies for each main station line provided.
- (6) Refer to Section 24 of this Tariff for applicable charges.

Issued: DEC 29 1983 Effective: JAN 0 1 1984

.

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

RECEIVED

SEP 2 5 1989

31.4.2 Centrex II Premises Vehicle Service-(Continued)

- -- -- -

B. Secondary Location

MISSOURI **Public Service Commission**

	Monthly Rate(2)		-
First 200 Station Lines (RXR) (RX2) (RUV) (RX5)(6) <u>NRX++(5)</u>	Next 300 Station Lines (RXR) (RX2) <u>(RUV) (RX5)(6)</u> <u>NRX++(5)</u>	Over 500 Station Lines (RXR) (RX2) (RUV) (RX5)(6) <u>NRX++(5)</u>	<u>1</u>
\$4.80	\$4.00	\$2.80	
		Monthly <u>Rate</u>	CANCELED
Trunks, with rotary of	peration	(3)	JUN 811893 #/
Touch-Tone Operation charge for trunks) .		(7)	- 2 ndR.S.
Minimum charge, per s	ystem	(4)	المالية الم

- (1) Refer to Paragraph 31.1.27 for obsolescence regulation.
- (2) Refer to Section 13 of this Tariff for applicable charges for DID and AIOD Services.
- (3) Refer to the Local Exchange Tariff for applicable PBX Trunk rates.
- (4) The minimum charge for Centrex station lines served by switching equipment located on customer's premises, excluding extension station lines and any other chargeable items of equipment or service, per Centrex System, shall be the rate applicable to 200 main Centrex station lines at the primary location and 50 main Centrex station lines at each secondary location.
- (5) The following NRX codes are applicable to Centrex C.U. secondary locations:

Effective:

		1st	Next	0ver	
		200	<u>300</u>	500	
	Same Serving Office	NRXYD	NRXYE	NRXYF	
	Contiguous Serving Office	NRXYG	NRXYH	NRXYJ	
	Noncontiguous Serving Office	NRXYK	NRXYL	NRXYM	
51	A Service and Equipment Charge of	\$28 50 ann	lies for	onch main	

(CR) (6) A Service and Equipment Charge of \$28.50 applies for each main station line provided.

(7) Refer to Section 24 of this Tariff for applicable charges.

FILED

OCT 1 1989

Issued: JEP 25 1900

By R. D. BARRON, President-Missouri Division Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

General Exchange Tariff

KEGEU

Section 31

1211

riginal Sheet-12

DEC 2 9 1003

UJIZZONKI

Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

31.4.2 Centrex II Premises Vehicle Service-(Continued)

B. Secondary Location

Monthly Rate(2) First 200 Next 300 **Over** 500 Station Lines Station Lines Station Lines (RXR) (RX2)(RXR) (RX2)(RXR) (RX2)(RUV) (RX5)(6) (RUV) (RX5)(6) (RUV) (RX5)(6) MRX++(5)MRX++(5)NRX++(5)\$4.80 \$4.00 \$2.80 Monthly Rate CANCELLED Trunks, with rotary operation. . . (3) OCT 1 1989 Touch-Tone Operation (in addition to BY MARS #12 charge for trunks) . . . (7) Public Service Cotamission MISSULA Minimum charge, per system (4) (1) Refer to Paragraph 31.1.27 for obsolescence regulation. (2) Refer to Section 13 of this Tariff for applicable charges for DID and AIOD Services. (3) Refer to the Local Exchange Tariff for applicable PBX Trunk rates.

(4) The minimum charge for Centrex station lines served by switching equipment located on customer's premises, excluding extension station lines and any other chargeable items of equipment or service, per Centrex System, shall be the rate applicable to 200 main Centrex station lines at the primary location and 50 main Centrex station lines at each secondary location.

(5) The following NRX codes are applicable to Centrex C.U. secondary locations: 1st Next Over FILED 200 300 5'00 NRXYF JAN - 1 1934 Same Serving Office NRXYD NRXYE NRXYJ 83 Contiguous Serving Office NRXYG NRXYH Noncontiguous Serving Office NRXYK NRXYL Con line'ssion (6) A Service and Equipment Charge of \$4.75 applies for each ma provided.

(7) Refer to Section 24 of this Tariff for applicable charges.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 31 1st Revised Sheet 13 Replacing Original Sheet 13

OBSOLETE SWITCHING SYSTEMS

RECEIVED SEP 2 5 1989

31.4.2 Centrex II Premises Vehicle Service-(Continued)

MISSOURI Public Service Commission

C. Service for Educational Institutions

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

Centrex II Service for dormitories administered by the educational institution may be provided at the following rates when associated with a Centrex System as provided by this Tariff.

Main Station lines

Monthly Rate(2) (PSA) (RV3)(3)(5) (PS7) (RTX)(4)(5) \$11.55 \$10.70

CANCELLED

Public Cervice Commission

0CI I 1909

89 - 14 Public Service Commission

(1) Refer to Paragraph 31.1.27 for obsolescence regulation.

(2) The customer shall elect the method in which billing is to be administered, and all dormitory station lines shall be billed accordingly.

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St Louis Missouri

Effective: OCT 01 1989

(3) The Telephone Company will bill and collect toll charges.

(4) The Institution will bill and collect all charges.

SEP 25 199

Issued:

(CR) (5) A Service and Equipment Charge of 28.50 applies for each main station line provided.

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.4 RATES - CENTREX C.U. SERVICE(1)-(Continued)

31.4.2 Centrex II Premises Vehicle Service-(Continued)

C. Service for Educational Institutions

Centrex II Service for dormitories administered by the educational institution may be provided at the following rates when associated with a Centrex System as provided by this Tariff.

Main Station lines

Monthly Rate(2) (PSA) (RV3)(3)(5) (PS7) (RTX)(4)(5)

\$11.55

\$10.70

CANCELLED

OCT 1 1989 BY AR S. #13 Public Service Commission MISSOURI

FILED

(1) Refer to Paragraph 31.1.27 for obsolescence regulation.

(2) The customer shall elect the method in which billing is to be administered, and all dormitory station lines shall be billed accordingly. 83-253

- (3) The Telephone Company will bill and collect toll charges. Public Service Commission
- (4) The Institution will bill and collect all charges.
- (5) A Service and Equipment Charge of \$4.75 applies for each main station line provided.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984



General Exchange-Tariff DECES

OBSOLETE SWITCHING SYSTEMS

31.5 CENTREX C.U. SERVICE - OFF-PREMISES STATIONS LINES - REGULATIONS SOURT

- Public Service Commission
- 31.5.1 Off-premises station lines are furnished subject to the availability of facilities and provided no unusual expense is involved.
- 31.5.2 An off-premises Centrex station line will be furnished the same service as a primary Centrex station line where equipment and facilities permit. When additional equipment or facilities are required, they may be provided under applicable sections of this and other Tariffs or on a Special Service Arrangement Request basis.
- 31.5.3 Off-premises station lines may be located on other premises of the customer or on the premises of someone other than the customer. The private line service charge treatment given in Section 2 of the Private Line Service Tariff applies in connection with such station lines.
- 31.5.4 With respect to the use of off-premises station lines, the following regulations apply:

Off-premises stations located on other premises (except different premises in the same building where standard operation and transmission can be obtained on all connections) are furnished for communication with other stations directly connected to the Centrex System, trunks, tie lines or other off-premises station lines, only subject to the condition that in some cases standard operation and transmission may not be obtained when such connections are established.

Off-premises stations located on a premises other than that of the customer are furnished with the understanding that such stations are provided only if some separate class of exchange service is also furnished at the off-premises location.

CANCELLED JUN 811993 #1 BY 2 mar. S. #1 rublic Service Contraiscion JAN - 1 1984 83-253 Public Service Commission

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 2nd Revised Sheet 15

Replacing 1st Revised Sheet 15

OBSOLETE SWITCHING SYSTEMS

RECEIVED

31.6 CENTREX C.U. SERVICE - FACILITY TERMINATIONS

31.6.1 Tie Lines

SEP 2.5 1989 MISSOURI

UN 211823 #

89 -

- A. A tie line is a facility directly connecting two PBXutGentrexior GimpuleSion systems. Tie lines are furnished subject to the availability of facilities and are provided when no unusual expense is involved.
- в. Tie lines may terminate in Centrex Systems of the same or different customers and on the same or different premises. For applicable private line charges, refer to Section 2 of the Private Line Service Tariff.
- c. The following regulations apply in connection with the use of tie lines associated with the Centrex Systems.
 - 1. Tie lines connecting Centrex Systems, or a Centrex System with a PBX or similar system, are arranged at the Centrex System end for standard operation and transmission when connected with trunks, off-premises station lines or other tie lines.
 - 2. Tie lines connecting Centrex Systems, or a Centrex System with a PBX or similar system, of different customers are provided only if each of the systems involved is connected to the serving central officer by trunks.
 - 3. Universal Tie Line Terminal Charges

The following Rates apply for a universal tie line terminal when asso-Typlic Cervice Oc.mmiccic iated with a Centrex System.(2)(3)

	Monthly <u>Rate</u>	Complex Installation/ Move Charge	Service and Equip- ment Charge
With operator and/or central office access, each (TLL)(1) Without operator or central		\$53.25	(CR)\$6.75
office access, each (TLN)(4)	69.45	53.25	(CR) 6.75

- (1) This service provides the facilities necessary within the Centrex System for a tie line termination and for connection to the Telecommunications Network.
- (2) Charges equivalent to the charges for these Universal Tie Line Terminals will apply for the termination of station lines of a Centrex System in attendants' positions and/or switching equipment of another Centrex System or a PBX System. The Telephone Company shall not be responsible for the through transmission of signals or of supervision when facilities are arranged in this manner.

(3) Not intended for use in terminating interstate Enhanced Private Switched Communications Service (EPSCS) channels, Type A, for customers served by No. 101 ESS Systems. 0CT 1

(4) Available where facilities permit.

Effective: UCI VI 1999 IC Service Commission Issued: SEP 25 1999 R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

31.6.1 Tie Lines

31.6 CENTREX C.U. SERVICE - FACILITY TERMINATIONS

General Exchange Tariff Section 31 1st Revised Sheet 15 Replacing Original Sheet 15

OBSOLETE SWITCHING SYSTEMS

5		REGEIVED
---	--	----------

- JUN 2 7 1956 A. A tie line is a facility directly connecting two PBX, Centrex or similar systems. Tie lines are furnished subject to the availability of and are provided when no unusual expense is involved. Public Service Commission
- Tie lines may terminate in Centrex Systems of the same or different cus Β. tomers and on the same or different premises. For applicable private line charges, refer to Section 2 of the Private Line Service Tariff.
- С. The following regulations apply in connection with the use of tie lines associated with the Centrex Systems.
 - 1. Tie lines connecting Centrex Systems, or a Centrex System with a PBX or similar system, are arranged at the Centrex System end for standard operation and transmission when connected with trunks, off-premises station lines or other tie lines.
 - Tie lines connecting Centrex Systems ANGELLED 2. or similar system, of different customers are provided only if each of the systems involved is connected to the serving central office by Why SHE D trunks.

Public Service Commission 3. Universal Tie Line Terminal Charges

The following Rates apply for a universal tie line terminal when assoiated with a Centrex System.(2)(3)

	Monthly Rate	Complex Installation/ Nove Charge	Service and Equipment Charge
With operator and/or central office access, each (TLL)(1) Without operator or central	\$77.00	\$53.2 5	(CR)\$3.75
office access, each (TLN)(4)	69.45	53.25	(CR) 3.75

- (1) This service provides the facilities necessary within the Centrex System for a tie line termination and for connection to the Telecommunications Network.
- (2) Charges equivalent to the charges for these Universal Tie Line Terminals will apply for the termination of station lines of a Centrex System in attendants' positions and/or switching equipment of another Centrex System or a PBX System. The Telephone Company shall not be responsible for the through transmission of signals or of supervision when facilities are arranged in this manner.

(3) Not intended for use in terminating interstate Enhanced Private Switched Communications Service (EPSCS) channels, Type A, for customers served No. 101 ESS Systems. Public Service Commission

(4) Available where facilities permit.

Issued: JUN 27 1936

Effective: JUL 1 1986

General Exchange Tariff

DE CSection 31 Original Sheet 15

DEC 2 9 1983

MISSOURI

Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.6 CENTREX C.U. SERVICE - FACILITY TERMINATIONS

31.6.1 Tie Lines

- A. A tie line is a facility directly connecting two PBX, Centrex or similar systems. Tie lines are furnished subject to the availability of facilities and are provided when no unusual expense is involved.
- Tie lines may terminate in Centrex Systems of the same or different cus-В. tomers and on the same or different premises. For applicable private line charges, refer to Section 2 of the Private Line Service Tariff.
- The following regulations apply in connection with the use of tie lines C. associated with the Centrex Systems.
 - Tie lines connecting Centrex Systems, or a Centrex System with a PBX 1. or similar system, are arranged at the Centrex System end for standard operation and transmission when connected with trunks, off-premises station lines or other tie lines.
 - 2. Tie lines connecting Centrex Systems, or a Centrex System with a PBX or similar system, of different customers are provided only if each of the systems involved is connected to the serving central office by trunks.
 - 3. Universal (Pie Pine Fermina p¹Charges

The following Rates apply for a universal tie line terminal when assoiated with a Centrex System. (2)(3)

BY ST RS. 5 PUBLIC SERVICE COMMISSION	Monthly Rate	Complex Installation/ Move Charge	Service and Equipment Charge
With operator and/or central office access, each (TLL)(1) Without operator or central	\$77.00	\$53.25	\$4.75
office access, each (TLN)(4)	69.45	53.25	4.75

- (1) This service provides the facilities necessary within the Centrex System for a tie line termination and for connection to the Telecommunications Network.
- (2) Charges equivalent to the charges for these Universal Tie Line Terminals will apply for the termination of station lines of a Centrex System in attendants' positions and/or switching equipment of another Centrex System or a PBX System. The Telephone Company shall not be responsible, for the through transmission of signals or of supervision when facilities are arranged in this manner.

(3) Not intended for use in terminating interstate Enhanced Private Switched Communications Service (EPSCS) channels, Type A, for customers serve 59 Public Service Commission No. 101 ESS Systems.

(4) Available where facilities permit.

Issued: DEC 29 1983 Effective: JAN 0 1 1984
No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.6 CENTREX C.U. SERVICE - FACILITY TERMINATIONS-(Continued)

31.6.2 Out-WATS Terminations

When an outward WATS line is terminated in the switching equipment of a Centrex System, the following Out-WATS Termination Charge is applicable. This termination provides dial access to WATS lines so equipped.

General Exchange Tariff

REGERENZIESheet 16

DEC 29 1023

MISSOURI Public Service Commission

JAN - 1 1984

Section 31

CANCELLER Complex Installation/ JUN 211993 <u>Rate</u> BY 2 M R. 5515.35 Move Charge Out-WATS Termination Charge, per WATS line (WLA). . . . (1)Public Gervice Commission 31.6.3 Network Access Terminals

- A. Network access terminals are furnished at Centrex Systems to provide station users access to the customer's SCAN- and CCSA-type networks.
- B. Dial network access terminals include the following service features: (2)
 - Network inward dialing of calls to station lines on the Centrex 1. System.
 - 2. Direct outward dialing of network calls from station lines of the Centrex System.
 - 3. Attendant transfer of incoming network calls from one station line of a Centrex System to another station line of that same system.
 - 4. Discrete identification to the attendant positions and stations on the SCAN- or CCSA-type network are furnished for communication between other attendant positions and stations on the same SCAN- or CCSA-type network.
- C. Access lines connecting attendant positions and stations on the SCAN- or CCSA-type network are furnished for community of CCSA-type network and stations on the same SCAN- or CCSA-type network and STATES AND A CCSA-type network are furnished for communication between other attendant
- (1) A Complex Installation/Move Charge of \$50.00 is applicable if installation? subsequent to the initial installation of the Centrex System Series Commission
- (2) These features are provided only where appropriate Centrex equipment is ---installed.

Issued: DEC 29 1983

Effective: JAN 0 1 1984

No supplement to this General Exchange Tariff tariff will be issued Section 31 except for the purpose 2nd Revised Sheet 17 Replacing 1st Revised Sheet 17 of canceling this tariff. **OBSOLETE SWITCHING SYSTEMS** RECEIVED 31.6 CENTREX C.U. SERVICE - FACILITY TERMINATIONS-(Continued) SEP 2 5 1989 31.6.3 Network Access Terminals-(Continued) MISSOURI D. Rates Public Service Commission 1. The following rates apply for a network access terminal in a Centrex System for use with a CCSA-type network: Complex Service Monthly Installation/ and Equip-Move Charge Rate ment Charge Dial network access terminals, each (901)(1). \$ 62.90 \$ 53.25 (CR)\$6.75 Manual termination in attendant position only, per network access 65.00 (CR) 6.75 2. The following rates apply for a network access terminal in a Centrex System for use with a SCAN network: Complex Service Monthly Installation/ and Equip-Rate Move Charge ment Charge Dial network terminals(1) With multilevel precedence preemption calling, each (DKB) . \$ 90.20 \$150.00 (CR)\$6.75 Without multilevel precedence 150.00 preemption calling, each (DKC) 83.00 6.75 Manual network access terminals Access lines will terminate Dig is u only in an attendant position with automatic network supervision and/or precedence pre-50.00 emption capabilities, each (BK1) 36.10 (CR) 6.75 FILED (1) This service provides the facilities necessary within the Centrex System for OCT 1 1989 CCSA-type network termination. 89 14 $-\Delta$ Issued: Effective: Public Service Commission SEP 25 199 Tict 0119 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

~ ...

-

-- -

- . .

Issued:	JUN 27 1986 Effect	ive: JUL	1 1986	
	service provides the facilities nec -type network termination.	27.51	Public Servi	ED 1986 System for ce Commission
	Access lines will terminate only in an attendant position with automatic network super- vision and/or precedence pre- emption capabilities, each (BK1)	BY	ervice Commiss MISSGUR	(CR) 3.75
	Manual network access terminals	CAN	VCELLED	5.75
	With multilevel precedence pre- emption calling, each (DKB) . Without multilevel precedence preemption calling, each (DKC)	\$ 90.20 83.00	\$150.00	(CR)\$3.75
	Dial network terminals(1)	Monthly <u>Rate</u>	Complex Installation/ Move Charge	Service and Equipment Charge
2	. The following rates apply for a new System for use with a SCAN network		ess terminal in	a Centrex
	Manual termination in attendant position only, per network access terminal (9CS)	32.40	65.00	(CR) 3.75
	Dial network access terminals, each (901)(1)	\$ 62.90	\$ 53.25	(CR)\$3.75
	· · ·	Monthly <u>Rate</u>	Complex Installation/ Move Charge	Service and Equipment Charge
1.	. The following rates apply for a no System for use with a CCSA-type no	etwork acc etwork:	ess te Piblic Se	
D. Ra		,	Į.	27 1986
	Network Access Terminals-(Continued			
1 6	OBSOLETE SWITCHING			BEIVED
	or the purpose ing this tariff.		lst Re Replacing Ori	vised Sheet ginal Sheet
	.ll be issued			xchange Tari: Section (

P	S	С	Mo	-No.	35

)	No supplement to this tariff will be issued except for the purpose of canceling this tariff.		General Exchange_Tariff DECESSECTION 31 DECESSECTION 31 DECESSECTION 31 DECESSECTION 31
		OBSOLETE SWITCHING SYSTEMS	DEC 25 1003
	31.6 CENTREX C.U. SERVICE	- FACILITY TERMINATIONS-(Continu	ed) 1.115500Kl
	31.6.3 Network Access Te	erminals-(Continued)	Public Service Commission
	D. Rates		

1. The following rates apply for a network access terminal in a Centrex System for use with a CCSA-type network:

	Monthly Rate	Complex Installation/ _Move Charge_	Service and Equipment <u>Charge</u>
Dial network access terminals, each (901)(1)	\$ 62.90	\$ 53.25	\$4.75
Manual termination in attendant position only, per network access terminal (9CS)	32.40	65.00	4.73

2. The following rates apply for a network access terminal in a Centrex System for use with a SCAN network:

	Dial network terminals(1)	Monthly Rate	Complex Installation/ Move Charge	Service and Equipment Charge
	With multilevel precedence pre- emption calling, each (DKB) . Without multilevel precedence preemption calling, each (DKC)	\$ 90.20	\$150.00 6官 L L馬D	\$4.75 4.75
	. Manual network access terminals	JUL	1 1986	
	Access lines will terminate only in an attendant position with automatic network super- vision and/or precedence pre-	BY 12	RS.17 RVICE COMMISSION OF MISSOURI	
	emption capabilities, each (BK1)	36.10	50.09	民团 4.75
			JAN -	1 198/)
(1) This CCSA	service provides the facilities nec- -type network termination.	essary wit		- 253 Sostemissfor
Issued:	DEC 2 9 1983 Effect	ive: JAI	N O 1 1984	

OBSOLETE SWITCHING SYSTEMS

- 31.7.1 Centrex service will be furnished at the same location under the commission schedules and conditions set forth below:
 - A. Primary Location

Primary locations will require a termination contract based on the contract schedule.

B. Secondary Location

Secondary locations will require a separate termination contract based on the contract schedule with the workable station line count starting from zero.

C. Contract Schedule(1)

Workable station line capacity

1. Primary Location

Switching equipment located on customer's premises	
First 200 station lines	\$82,950
Next 300 station lines, per 100	
station lines	34,750
Each additional 100 station lines	20,650

CANCELLED JUN 211993 #, BY 2 nd RS. #, Public Cervice Commission FILED JAN - 1 1934 83 - 253 Public Service Commission

Termination Charges

General Exchange Tariff

RE RSection 31

Original5Shee 6 18

DEC 20 1983

(1) Contract term is three years.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

General Exchange Tariff No supplement to this tariff will be issued except for the purpose Original Sheet 19 of canceling this tariff. DEC 2 9 1983 OBSOLETE SWITCHING SYSTEMS MISSOURI 31.7 CENTREX C.U. SERVICE - TERMINATION CHARGES AND CONTRACTUDER SEASE 31.7.1 (Continued) C. Contract Schedule(1)-(Continued) CANCELLED Workable station line capacity-(continued) JUN 211993 #/ Secondary location 2. Public Service Co.achtelua Switching equipment located on а. customer's premises Termination Charges (1) Located in same central office area as primary location First 200 station lines, per 50 station lines. \$19,900 Next 300 station lines, per 17,800 50 station lines. Each additional 100 station lines 20,650 (2) Located in contiguous central office area to primary location First 200 station lines, per 50 station lines. 23,600 Next 300 station lines, per 50 station lines. 19,900 Each additional 100 station lines 24,450 (3) Located in noncontiguous central office area to primary location First 200 station lines, per 23,550 50 station lines. Next 300 station lines, per 17111 21,850 50 station lines. Each additional 100 station JAN 28,300 1 1984 lines 83 - 253 Public Service Commission

(1) Contract term is three years.

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.



OBSOLETE SWITCHING SYSTEMS

31.7 CENTREX C.U. SERVICE - TERMINATION CHARGES AND CONTRACT TERM-(Continued)-

31.7.1 (Continued)

- D. Contract Conditions
 - 1. Termination charges will apply for Centrex Service if discontinued or transferred to another address before the expiration of the initial contract period. Termination charges as set forth above shall be reduced by 1/36 for each month the service is retained in service at the same location.
 - 2. Initial termination contract amounts are based on the one-year forecast of the workable station line capacity. Separate termination contracts are required as additional working station lines are connected and when the workable station line capacity exceeds the range of the present contract schedule. These termination contracts will be executed for the difference between the existing termination contract amount and the amount specified by the new contract schedule. Contracts will run concurrently until expired or terminated.
 - 3. Termination charges will apply when a Centrex System is reduced to the extent that the workable station line capacity falls into the next lower range of the contract schedule and equipment is removed.
 - 4. When facilities directly associated with the provision of Centrex Service are disconnected, which are covered by a Contract Term, applicable charges are determined on a last-in first-out basis.

CANCELLED JUN 211393 BY 2 na R. rublic Cervice Commission FILED

JAN = 1 1934 8 3 - 2 5 3 Public Service Convertation

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

CENTREX C.U. SERVICE - TERMINATION CHARGES AND CONTRACT TERMENCONTRINED mmission

Original Sheet 21

DEC 2 9 1003

General Exchange Wahiff

31.7.1 (Continued)

31.7

D. Contract Conditions-(Continued)

5. Cancellation of Contract

Where the applicant cancels an application for service prior to start of engineering, manufacturing or installation work, no charge applies.

When engineering, manufacturing or installation of facilities has been started prior to the cancellation, the charge specified below, whichever is lower, applies.

A charge equal to the estimated costs incurred in connection with such engineering, manufacturing or installation work, less estimated net salvage.

The charge for the minimum period of the service ordered by the customer as provided in this Tariff plus the full amount of any termination charges applicable.

6. The move of attendant consoles from the current primary location to a secondary location, thus making that secondary a new primary location by definition, does not constitute a termination of service. Charges for this type of move will be based upon cost.

CANCELLED JUN 211993 #1 reablic Service Commission



Issued:

d: DEC 2 9 1983

Effective: JAN 0 1 1984

....

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 31 2nd Revised Sheet 22 Replacing 1st Revised Sheet 22 RECEIVED

OBSOLETE SWITCHING SYSTEMS

31.8 CENTREX G.U. SERVICE

MISSOURI

SEP 25 1989

31.8.1 Centrex C.U. Optional Features

21.0.1	Centrex C.U. Optional Features			Masouri Conico Con	amission
			Public Service Comm		innourou.
		Monthly	Complex Installatio	Servi	
		Rate	Move Charg		
		Nate	Hove charg	e ment Cha	rke
	Additional Attendant's Position, Console or Switchboard(1)	\$383.	00	(())))	15
	(RXXWC) (RXYWC)			• (CR)\$6	.23
	equipped or fraction thereof (EBR) Attendant Conference Arrangement(3)		15	• 6	.75
	(RKT)		50 \$250.	00 6	5.75
	Per group of 20 lamps, per posiition (PQ5)	7.	60	. 6	5.25
	Cable Turning Section each, (CQM)(11)(12)	8.	60 50.	.00 6	5.75
	Per group of 100 station lines or fraction thereof (E6GSY) Call Forwarding Don't Answer(5)(6)	19.	20 300.	00 6	5.75
	Per group of 100 station lines or fraction thereof (E9GSY)	. 28.	75 350.	.00 6	5.75
	Call Forwarding - Variable Per main station line equipped (EA Call Hold and Call Pickup	T) 1.	60 1	.00(7)	5.75
	Per main station line equipped(8) (EA2)		.60 1.	.00(7)	5.75
	Call Transfer - Individual - All Ca Consultation Hold - All Calls and T Way Calling	hree-			
	Per 100 main station lines equippe fraction thereof (E2H)(5)	46.	807		6.75
	Code Call Access Arrangement (CCX) Conference Calling Station Line	RICEAS	40 300		6.75
	. Arrangement (EAN)(10) ن	UN 230	150 ± 150	.00 (CR)	6.75
	Б	UN 2305 12 ~ 1 Convice	Commission	FILED	
(See Shee	La Polonic	35101-		OCT 1 1989	9
	t 23 and 24 for Footnotes.)		Public	Service Com	mission

Issued: SEP 25 1900

Effective:OCT 01 1980

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 1st Revised Sheet 22

- ---

Replacing Original Sheet 22

OBSOLETE SWITCHING SYSTEMS

Monthly Rate

31.8 CENTREX C.U. SERVICE

REGEIVED Complex 27 Eservice and Installation/ Equipment Move Charge

31.8.1 Centrex C.U. Optional Features

		THOVE CHADGEOUL		
		Public Service C	ommission	
Additional Attendant's Position, Console or Switchboard(1) (RXXWC) (RXYWC)	\$383.90		(CR)\$3.75	
Attendant Camp-On with Indication(2) Per group of 100 station lines	4909.70	,		
equipped or fraction thereof (EBR) Attendant Conference Arrangement(3)	67.15		3.75	
(RKT)	57.50	\$250.00	3.75	
Per group of 20 lamps,				
per posiition (PQ5)	7.60		3.75	
each, (CQM)(11)(12)	8.60	50.00	3.75	
Per group of 100 station lines				
or fraction thereof (E6GSY)	19.20	300.00	3.75	
Call Forwarding Don't Answer(5)(6)				
Per group of 100 station lines or				
fraction thereof (E9GSY)	28.75	350.00	(CR) 3.75	
Call Forwarding - Variable				
Per main station line equipped (EAT)	1.60	1.00(7)	3.75	
Call Hold and Call Pickup				
Per main station line equipped(8)				
(EA2)	1.60	1.00(7)	3.75	
Call Transfer - Individual - All Call				
Consultation Hold - All Calls and Thr	ee-			
Way Calling				
Per 100 main station lines equipped			0 7 5	
fraction thereof $(E2H)(5)$	46.80	200 00	3.75	
Code Call Access Arrangement (CCX)(9)		300.00	3.75	
Conference Calling Station Line Arrangement (EAN)(10)	3005	150.00	3.75	
- a NICE		Contraction of the second s		
CAUSE	1989 _	FIL	ED)	
and 300	1989 R.S.#22	ission WL 1	1986	
(See Sheet 23 and 24 for Footnotes.)	SSOURI	86- Public Service	84 Commission	
Issued: JUN 27 1986 Effectiv	14.11	1 1986	•••••••••	
	(icon	inician		
By R. D. BARRON, President-N Southwestern Bell Teler				
Southwestern Bell Telephone Company St. Louis Missouri				

St. Louis, Missouri

.....

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

Monthly

Rate

31.8 CENTREX C.U. SERVICE

٠

31.8.1 Centrex C.U. Optional Features

Ç	Reneral Exc	hange Tar Section nal/Sheet	
	DEC	2.6 (203	
I	e Mit	SSOUKI	1
	Public Servi	ice Commiss	sion
Insta	mplex allation/ e Charge	Service Equipme Charg	nt

	Additional Attendant's Position,			
	Console or Switchboard(1)			
	(RXXWC) (RXYWC)	\$383.90		\$4.75
	Attendant Camp-On with Indication(2)			
	Per group of 100 station lines			<i>.</i>
	equipped or fraction thereof (EBR)	67.15		4.75
	Attendant Conference Arrangement(3)			/
	(RKT)	57.50	\$250.00	4.75
	Busy Indication Lamp(4)			
	Per group of 20 lamps,	7 (0		/ 75
	per posiition (PQ5)	7.60		4.75
	Cable Turning Section	a (a		/ 75
	each, (CQM)(11)(12)	8.60	50.00	4.75
	Call Forwarding - Busy Line(5)(6)	-		
	Per group of 100 station lines			/ ==
	or fraction thereof (E6GSY)	19.20	300.00	4.75
	Call Forwarding Don't Answer(5)(6)			
	Per group of 100 station lines or	~~ ~~		/
	fraction thereof (E9GSY)	28.75	350.00	4.75
	Call Forwarding - Variable			
	Per main station line equipped (EAT)	1.60	1.00(7)	3.75
	Call Hold and Call Pickup			•
	Per main station line equipped(8)			
	(EA2)	1.60	1.00(7)	3.75
	Call Transfer - Individual - All Call	•		
	Consultation Hold - All Calls and Thr	ee-		
	Way Calling			
	Per 100 main station lines equipped			a
	fraction thereof (E2H)(5)	46.80		3.75
	Code Call Access Arrangement (CCX) (9)	R 1891495	<u>300.00</u>	3.75
	Conference Calling Station Life NU	15555		o
	Arrangement (EAN)(10)	30.75	150.00	3.75
	JUL	1 1986		
	JÓĽ			
	1 et 1	2.5.20	► ¦ [5]	」、「「」、「」、「」、「」、「」、「」、「」、「」、「」、「」、「」、「」、「
	BY	ICE COMMISS	ION	
	PUBLIC SERV	MISSOURI	ι	- 1 1984
(0. 0)			UMN	
(See She	et 23 and 24 for Footnotes.)		8	3 - 253
		<u> </u>	Public Cons	ica Commission
Issued:	DEC 29 1983 Effectiv	14.1	N 0 1 1984	
issueu:	DEC 2 9 1983 Effectiv	ve: JAI	Y V I 1304	

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.8 CENTREX C.U. SERVICE-(Continued)

31.8.1 Centrex C.U. Optional Features-(Continued)

Footnotes:

- A three-year termination contract is applicable in the amount of \$5,000.00 per position.
- (2) A complex Installation/Move Charge of \$50.00 per system will apply regardless of the number of station lines involved when installed subsequent to the initial installation of the system.
- (3) This is an attendant-controlled conference arrangement. This conferencing equipment provides for the simultaneous connection of several Centrex station lines on the same premises. Connection of lines, other than as stated above, is subject to the limitations of equipment; and where such equipment can be connected to Telephone Company Exchange Service, Private Line Service or Long Distance Message Telecommunications Service, the responsibility of the Telephone Company shall be limited to the furnishing of facilities suitable for Exchange Service, Private Line Service or Long Distance and operation of such facilities in a manner proper for such services.
- (4) A Complex Installation/Move Charge of \$200.00 per installation will apply regardless of the number of lamps provided when installed subsequent to the initial installation and when no other switchboard or dial equipment additions are being made at the same time.
- (5) This feature is available to all station lines in a system and is provided only on a system basis. One Complex Installation/Move Charge is applicable per system.
- (6) This feature will function only on incoming calls from the Telecommunications Network.
- (7) When Call Hold, Call Pickup and Call Forwarding-Variable are installed simultaneously, a Complex Installation/Move Charge of \$1.00 per main station line equipped will apply in lieu of the Complex Installation/Move Charge above.

CANCELLED JUN 211393 #, BY 2 7 4 R.S. FILED JAN - 1 1984 Public Service Commissio **83 - 253** Public Service Commission

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Section 31 Offerer 23 REGEVED DEC 2 9 1983 MISSOURI

Public Service Commission

General Exchange Tariff

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.8 CENTREX C.U. SERVICE-(Continued)

31.8.1 Centrex C.U. Optional Features-(Continued)

Footnotes:-(Continued)

- (8) These features are provided as a group per station line equipped; however, a station line must be equipped for Call Hold before it can be equipped for Call Pickup. When one station line in a "Dial Pickup Group" is equipped for Call Pickup, all station lines in that same group are automatically equipped with the same feature; therefore, the rate and Complex Installation/Move Charge will apply to each station line in the "Dial Pickup Group."
- (9) Regulations covering the provisions of this service are applicable as provided in Section 4 of this Tariff.
- (10) This is a station line controlled conferencing arrangement. This conferencing equipment provides for the simultaneous connection of several Centrex station lines on the same premises. Connection of lines other than as stated above is subject to the limitations of the equipment, and where such equipment can be connected to Telephone Company Exchange Service, Private Line Service or Long Distance Message Telecommunications Service, the responsibility of the Telephone Company shall be limited to the furnishing of facilities suitable for Exchange Service, Private Line Service or Long Distance Message Telecommunications Service and to the maintenance and operation of such facilities in a manner proper for such services.
- (11) Applies only to new installations of switchboard positions CANCELLED

JUN 211393 #, BY2 ~ R.S.

Public Service Commission

General Exchange Tariff

DEC 20 1000

MISSOURI

Public Service Commission I

REGEI

Original-Sheet 24

Section 31

C 10



Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

General Exchange Tariff Section 31 3rd Revised Sheet 25 Replacing 2nd Revised Sheet 25

OBSOLETE SWITCHING SYSTEMS

RECEIVED

31.8 CENTREX C.U. SERVICE-(Continued)

SEP 2 5 1999

31.8.1 Centrex C.U. Optional Features-(Continued)

		MILSS	2011
	Monthly <u>Rate</u>	ComplexService Installation/ Move Charge	COServiceOf and Equip- ment Charge
Dictation Access Arrangement (RC8)(1)(2)(3)(4)	\$ 38.40	\$200.00	
Paging Access Arrangement			
(55N)(1)(4)(5)	28.80	150.00	
Attendant Position(s) and first unrestricted main station lines	25		
equipped (E2G)(7)	46.80	50.00(8)	(CR)6.75
For each additional main station equipped (E2G12)(7) Supervisory Cabinet, each (LNR)(1.95	1.00(8) 50.00	6.75 (CR)6.75

CANCELLED

JUN 211993 BY 2nd R.S. #1 Public Service Commission

(See Page 26 for Footnotes.) FILED OCT 01 1990 $0071 \frac{1}{989}$ Effective: Issued: SEP 25 1990 By R. D. BARRON, President-Missouri Divisiuplic Service Commussion Southwestern Bell Telephone Company St Iouis Missouri

No supplement to this General Exchange Tariff tariff will be issued Section 31 except for the purpose 2nd Revised Sheet 25 of canceling this tariff. Replacing 1st Revised Sheet 25 OBSOLETE SWITCHING SYSTEMS REGEU 31.8 CENTREX C.U. SERVICE-(Continued) JUN 2 7 1986 31.8.1 Centrex C.U. Optional Features-(Continued) Complex Source and stallation Equipment Monthly HUDIE CEAUge Commission Rate Dictation Access Arrangement (RC8)(1)(2)(3)(4)....\$ 38.40 \$200.00 Paging Access Arrangement $(55N)(1)(4)(5) \dots$ 28.80 150.00 Speed Calling(6) Attendant Position(s) and first 25 unrestricted main station lines equipped (E2G)(7). 46.80 50.00(8) (CR) 3.75 For each additional main station line equipped (E2G12)(7). . . . 1.00(8)3.75 1.95 Supervisory Cabinet, each (LNR)(9). 50.00 19.20 (CR) 3.75

CANCELLED OCT_1 1989 Public Service Commission MISSOURI FILED (See Page 26 for Footnotes.) 1986 JUL 1 86-84 Issued: JUN 27 1986 Effective: JUL Public Service Commussion By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

	No supplement to this tariff will be issued except for the purpose of canceling this tariff.	General Exchange Tariff Section 31 lst Revised Sheet 25 Replacing Original Sheet 25
	OBSOLETE SWITCHING SYSTEMS	REGEIVED
	31.8 CENTREX C.U. SERVICE-(Continued)	
	31.8.1 Centrex C.U. Optional Features-(Continued)	NOV 3 0 1994
	. Month Rate	
(RT)	Dictation Access Arrangement (RC8)(1)(2)(3)(4)\$ 38.	40 \$200.00
()	Paging Access Arrangement (55N)(1)(4)(5)	80 150.00
	unrestricted main station lines equipped (E2G)(7)	80 50.00(8) 4.75 95 1.00(8) 4.75
	Supervisory Cabinet, each (LNR) (9). 19.	. ,

GANGELLED

JUL 1 1986

BY 2 CHAR R.S. 25 PUBLIC SERVICE COMMISSION OF MISSOUNI



No supplement to this tariff will be issued except for the purpose of canceling this tariff.		General Exch	hange Tariff <u>Section 31</u> halvSheet 25
OBSOLETE SWITCHING S	YSTEMS	DEC 2	0 (223
31.8 CENTREX C.U. SERVICE-(Continued)		21	
31.8.1 Centrex C.U. Optional Features-(Conti	nued)	n	SUURI e Commission
	Monthly <u>Rate</u>	Complex Installation/ Move Charge	Service and Equipment Charge
Dictation Access Arrangement (RC8)(1)(2)(3)(4) Emergency Reporting Dial PBX	\$ 38.40	, \$200.00	
Station $(E4P)(10)(11)$.	3.90	25.00	\$4.75
Paging Access Arrangement			
<pre>(55N)(1)(4)(5)</pre>	28.80	150.00	

equipped (E2G12)(7).

Supervisory Cabinet, each (LNR)(9).



46.80

1.95

19.20

50.00(8)

1.00(8)

50.00

4.75

4.75

4.75

同旧国间 JAN = 1 1934 83 - 253 nnin. Public Service Comm

(See Page 26 for Footnotes.)

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

31.8 CENTREX C.U. SERVICE-(Continued)

31.8.1 Centrex C.U. Optional Features-(Continued)

Footnotes:

- (1) Regulations covering the provisions of this service are applicable as provided in Section 4 of this Tariff.
- (2) An attendant's telephone (DCF) station line, when required, may be provided at the Centrex extension line rate.
- (3) Includes 75 feet of multiconductor wiring between the dial dictation recording terminal equipment and the recording equipment.
- (4) Obsolete-applicable to existing installations at existing locations for existing customers.
- (5) Includes attendant's paging key and answer arrangement when required.
- (6) Applies to Centrex C.U. Service provided by 101 ESS.
- (7) A Service Charge of \$10.00 applies for each group of ten Speed Calling Numbers or fraction thereof that are simultaneously added, removed or changed subsequent to the initial installation of the Speed Calling feature.
- (8) Establishes the initial repertory.
- (9) The first cabinet is included as a standard feature of the Centrex Service when the customer has three or more attendant's console positions. Each cabinet over one is provided at the Tariff rate.

(RT)



JUN 211993 #

Public Service Commission

			FILED
Issued:	NOV 30 1984	Effective: DEC 31 1984	DEC 3 1 1984
	Ву	R. D. BARRON, President-Missouri Divis: Southwestern Bell Telephone Company St. Louis, Missouri	onuite Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 31 Offginal Sheet 26 MISSUUKI Public Service Commission

JAN = 1 1937 **8 3 - 2 5 3** Public Service Commission

OBSOLETE SWITCHING SYSTEMS

31.8 CENTREX C.U. SERVICE-(Continued)

31.8.1 Centrex C.U. Optional Features-(Continued)

Footnotes:

Issued:

DEC 2 9 1983

- Regulations covering the provisions of this service are applicable as provided in Section 4 of this Tariff.
- (2) An attendant's telephone (DCF) station line, when required, may be provided at the Centrex extension line rate.
- (3) Includes 75 feet of multiconductor wiring between the dial dictation recording terminal equipment and the recording equipment.
- (4) Obsolete-applicable to existing installations at existing locations for existing customers.
- (5) Includes attendant's paging key and answer arrangement when required.
- (6) Applies to Centrex C.U. Service provided by 101 ESS.
- (7) A Service Charge of \$10.00 applies for each group of ten Speed Calling Numbers or fraction thereof that are simultaneously added, removed or changed subsequent to the initial installation of the Speed Calling feature.
- (8) Establishes the initial repertory.
- (9) The first cabinet is included as a standard feature of the Centrex Service when the customer has three or more attendant's console positions. Each cabinet over one is provided at the Tariff rate.
- (10) Applies only to switchboard positions.
- (11) In addition, the appropriate station line rate will apply.

Effective:

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

CANDELLED DEC 31 1984 DE RS 200 BY LIC SERVICE CONNISSION PUBLIC SERVICE CONNISSION FILLED

JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(RT) 31.9

General Exchange Tariff Section 31 2nd Revised Sheet 27 Replacing 1st Revised Sheet 27 1st Revised Sheet 28 1st Revised Sheet 29 Original Sheet 30 through OBSOLETE SWITCHING SYSTEMS JUN 2 7 1980 MISSUURI Public Service Commission

CANCELLED JUN 211993 #1 Public Service Commission



Issued: JUN 27 1985

Effective: JUL 1 1986

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Genera	1 Exchang	ge Tari	lff
	Se	ection	31
lst	Revised	Sheet	27
Replacing	Qriginal_	Sheet	27
MS		GEI	

OBSOLETE SWITCHING SYSTEMS

(CP) 31.9 COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS

- 31.9.1 General Regulations
 - A. This Tariff is applicable to Obsolete Communications Service Package Premises Systems served by #101ESS switching vehicles that containing Mission to be maintained by the Telephone Company.
 - B. Series 100, 200 and 300 Communications Service Package Premises Systemsthe Telephone Company will determine and provide.

The attendant positions, as required for exchange and other services, will be determined by the Telephone Company. When customers request additional attendant positions beyond those required or determined by the Telephone Company, charges for such positions will be applied as set forth in accordance with the provisions of the General Exchange Tariff. The Telephone Company reserves the right to remove switching equipment, attendant positions and other facilities furnished by the Telephone Company when, in its judgment, such equipment or facilities are no longer required or necessary to provide exchange or other services furnished by the Telephone Company. Provision of optional features is limited to the capabilities of the serving vehicle. See Paragraph 31.9.6 for additional charges applicable to Communications Service Package Premises Systems.

- C. Additional equipment or facilities required as a result of seasonal business requirements or for customer-owned facilities will be charged for on an individual-case basis.
- D. The following terms is used Find this Section shall mean:

Primary Location - The customer's premises where the attendant equipment is located and where calls to the main listing are completed.

Secondary Location The customer's premises other than the location where the attendative customerselous located. OF MISSOURI Workable Stations - Those station lines available for customer use.

- E. All operating at customers' premises is performed at the expense of the customer and must conform to whatever rules and regulations the Company may consider necessary to maintain a proper standard of service.
- F. Other facilities, miscellaneous and supplemental equipment requested by customers and not detrimental to this service or other services of, the Telephone Company, will be furnished in accordance with regulations and at the rates specified in the applicable section of this Tariff.

Issued: NOV 3 0 1984

Effective: DEC 3 1 1984 DEC 3 1 1984

No supplement to this General tariff will be issued except for the purpose anceling this tariff.	1 Exchange Taviff Desection 31 Original Sheet 27 DEC 20 1003
OBSOLETE SWITCHING SYSTEMS	MISSOURI
9 COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS - FEATURES	Public Service Commission

- 1- 7 - --

Complex

A . 3 7 . A * . . /

31.9.1 Custom Calling Service

Rate	Move Charge
\$46.80	\$50.00
	Rate

下几上三① JAN - 1 1934 83 - 253 Public Service Commission

Service and

Equipment Charge

\$4.75

GANDELLED DEC 3 1 1984

PUBLIC SERVICE COMMISSION

A Service Charge of \$10.00 applies for each group of ten Speed Calling Numbers or fraction thereof that are simultaneously added, removed or changed subsequent to the initial installation of the Speed Calling feature.

Issued:

DEC 2 9 1983

Effect

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 1st Revised Sheet 28 Replacing Original Sheet 28

KEUEIN

MISSOURI

1210

OBSOLETE SWITCHING SYSTEMS

(CP) 31.9 COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS-(Continued) NOV 30 (S8)

31.9.1 General Regulations-(Continued)

- G. The necessary floor space, power wiring, power outle Public Sector Commission power for the operation of Private Branch Exchange Service are furnished by the customer, who assumes all responsibility for the safe condition of such floor space, wiring, outlets and power.
- H. The primary location of a customer served by switching equipment located on his premises must be in the central office area from which his listed telephone number is furnished.
- I. Service furnished at secondary locations must be of the same grade and class of service as that furnished at the primary location.
- J. A secondary Communications Service Package Premises System location is furnished the same calling scope as that furnished at the primary location.
- K. Service furnished at a secondary location will be provided under the same provisions of service as that furnished at the primary location.
- L. Secondary locations of a Communications Service Package Premises System will be furnished only if located in the same exchange as the primary location or within exchanges or zones of the Metropolitan Area.
- M. When a customer requests special equipment or modifications of standard equipment, it will be furnished whenever possible if not detrimental to any of the services or equipment of the Telephone Company. Such special equipment or modification of standard equipment will be furnished at additional charges.
- N. Mileage charges as specified in the Private Line Service Tariff apply to station lines extending outside the same building or beyond the same premises at both primary and secondary locations. Urban mileage charges for facilities extending beyond the base rate area apply at a primary location as specified in the Private Line Service Tariff. Such charges do not apply to secondary locations.
- 0. Universal Tie Line Terminal rate And Daply GE each tie line terminating in a Communications Service Package Premises System (primary or secondary). JUL 1 1986
- P. Switched Service Network Terminal rates shall apply for each SCAN- and CCSA-type access lineyterminating Cin & Tommunications Service Package Premises SystempuBLIC SERVICE COMMISSION
- Q. Directory listings are furnished under the provision of the applicable sections of this Tariff.

Issued: NOV 30 1984	Effective: DEC 31 1982	FIILED
By R. D. BARRON,	President-Missouri Division	DEC 3 1 1934
	Bell Telephone Company Louis, Missouri	Public Service Commission
· · · · · · · · · · · · · · · · · · ·		Uttimission والكرية المالية المالية الم

EGE

General Exchange Tariff

1.15500KI

Public Service Commission (

Original Sheet 28

JAN - 1 1984 83 - 253

Section 31

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS - FEATURES-(Continued) 31.9

31.9.1 Custom Calling Service-(Continued)

	Monthly <u>Rate</u>	Complex Installation/ Move Charge	Service and Equipment Charge
Custom Calling Service Features-(Continued) Speed Calling- (Continued) For each additional main station line equipped(2) Call Hold and Call Pickup,	\$1.95	\$1.00(1)	\$4.75
per main station line equipped (EA2)(3) Call Forwarding - Variable, per main station line	1.60	1.00(4)	3.75
equipped (EAT)	1.60	1.00(4)	3.75
	GANDE		
	DEC 3	1 1984	
	int L	CE COMMISSION	FILED
· · · · ·	PUBLIC SERVI	MISSOURI	JAN - 1 1984

- (1) Establishes the initial repertory.
- (2) A Service Charge of \$10.00 applies for each group of ten Speeducalling Numbersission or fraction thereof that are simultaneously added, removed or = changed - subsequent. to the initial installation of the Speed Calling feature.
- (3) These features are provided as a group per station line equipped; however, a station line must be equipped for Call Hold before it can be equipped for Call Pickup. All station lines in that same group are automatically equipped with the same features; therefore, the above rate and Installation Charge will apply to each station line in the "Dial Pickup Group."
- (4) When Call Hold, Call Pickup and Call Forwarding-Variable are installed simutaneously, a Complex Installation/Move Charge of \$1.00 per main station line equipped will apply in lieu of the above Complex Installation/Move Charges.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

- - -----

- -----

· · · · · · · · · · · ·

.

	tarif excep	f wi ot fo	ment to this 11 be issued r the purpose ing this tariff.	ls	al Exchange Tariff Section 31 It Revised Sheet 29 Original Sheet 29
	((7) 31 0	റസ	MINICATIONS SERVICE P	OBSOLETE SWITCHING SYSTEMS	
			General Regulations-	ACKAGE PREMISES SYSTEMS-(Conti	
	F	R. N 1	onrecurring charges f n the Local Exchange	or service and equipment apply Tariff.	MISSOURI PauliseSeficeTCommission
)	5	S P c	ystem main station li BX main station line	for a Communications Service H ne provides for a PBX main sta consists of all facilities, in switching system to the deman	tion line. The cluding inter-
	1	а		e Package Premises Systems, Ser to existing installations at ex stomers:	-
				permitted after the initial 1 after November 26, 1979, only vehicle.	
			his obsolescence is a ocation.	applicable to each primary and	each secondary
	31	.9.2	Series 100(1)(2)(3)		
				d features are provided with th Package Premises System:	ne Series 100
		1	Station-to-Station Restriction from (n (console or switchboard), Din n Calling, Station Hunting, Cal Dutgoing Calls, Power Failure 7 ard rotary dial service.	11 Transfer - Attendant,
	(NR)	2	2. Rates - Primary	GANGELLED	Monthly
			Next 80 station 1: Over 100 station		Rates \$22.95 6.85 4.60 . 55
	(2) (3)	The m Stati arran	ninimum monthly charge lon Hunting is limited ngement. Requests fo	, T., for obsolescence regulate e is for 20 station lines, per d to not more than ten station r above ten consecutive station ngement Request basis.	location. s with each hunting
	Issu	ed:	NOV 3 0 1984	Effective: DEC 31	1984 DEC 3 1 1934
				RRON, President-Missouri Divis stern Bell Telephone Company St. Louis, Missouri	

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

OBSOLETE SWITCHING SYSTEMS

MISSOURI Public Service Commission

> Service and Equipment Charge

Ofiginal Sheet 29

50 change Tariff,

Section 31

General Exch

COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS - FEATURES-(Continued) 31.9

31.9.1 Custom Calling Service-(Continued)

		Complex
	Monthly Rate	Installation/ Move Charge
Custom Calling Service		•
Features-(Continued)	-	
Call Transfer -		•
Individual - All Calls,		
Consultation Hold - All		
Calls and Three-Way		
Calling(1)		
Per 100 main station		
lines equipped or		
fraction thereof		
(E2H)(2)	\$46.80	

\$4.75





(1) Available only to Series 300 installation.

(2) This feature is available to all station lines in a system and is provided only on a system basis.

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

	No supplement to this tariff will be issued		Exchange Tariff Section 31	
,	except for the purpose of canceling this tariff.	C	REGEIVED	
	OBS	OLETE SWITCHING SYSTEMS		
(CP)	31.9 COMMUNICATIONS SERVICE P.	ACKAGE PREMISES SYSTEMS-(Continue	a) NOV 301984	
	31.9.3 Series 200(1)(2)(3)	(4) (5) (6)	MISSOURI Public Service Commission	

- The following standard features are provided with the Series-200. Α. Communications Service Package Premises System:
 - Attendant Position (console or switchboard), Direct Outward Calling, 1. Station-to-Station Calling, Station Hunting, Call Transfer - Attendant, Restriction from Outgoing Calls, Power Failure Transfer, Night Service, Attendant Direct Station Selection (with busy lamp field), Attendant Camp-On (with indication), Attendant Conference and Touch-Tone

(NR)

2. Rates - Primary

								Monthly <u>Rates</u>
1st 20 station lines, per line					•			\$24.90
Next 80 station lines, per line	•	•	•	•	•	•	•	7.20
Over 100 station lines, per line .	•		•				•	5.90
Extension station lines, per line.	•	•	•	•	•	•	•	.55

GANGELLED

JUL1 1986 SERVICE COMMISSION OF MISSOURI

- (1) Attendant Direct Station Selection with Busy Lamp Field is provided only with consoles and is applicable to first 200 station lines. Requests for this service beyond 200 station lines may be provided on a Special Service Arrangement Request basis, where facilities permit.
- (2) The minimum monthly charge is for 20 station lines per location.
- (3) Station Hunting is limited to not more than ten station lines with hunting arrangement. Requests for service above ten consecutive station lines may be provided on a Special Service Arrangement Request basis.
- (4) Refer to Paragraph 31.9.1, T., for obsolescence regulation.
- (5) Refer to the Local Exchange Tariff for applicable rates for PBX trunks.
- (6) Refer to Section 24 of this Tariff for applicable charges for Touch-Tone Service.

Effective: DEC 31 1984 DEC NOV 3 0 1984 Issued: By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

General Exchange Tariff Section 31 Original Sheet 31

OBSOLETE	SWITCHING	SYSTEMS
----------	-----------	---------

REGEDVED

NOV 3 0 1984

(CP) 31.9 COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS-(Continued)

31.9.4 Series 300(1)(2)(3)(4)(5)(6)(8)(9)

In addition to all of the standard features provided with the SOURI Series 200 System, the following additional standard field coesvice Commission are provided with a Series 300 System: Call Transfer - Individual, Consultation Hold, Add-On.

ates - Prímary	Monthly Rates
st 20 station lines, per line	\$26.65
ext 80 station lines, per line	9.15
ver 100 station lines, per line	7.35
stension station lines, per line	.55
	st 20 station lines, per line

^{31.9.5} Secondary Locations(4)(7)

Service will be limited to the provision that all standard and optional features cannot be provided at the secondary location. Primary rates will apply when providing secondary service. Tie line facilities required to connect the primary and secondary will be billed as specified in the Private Line Service Tariff.

.111 1986

(1)	Attendant Direct Station Selection by the Dusy Lamperterd is provided only with consoles and is applicable to first 200 station lines. Requests for
	with consoles and is applicable to first 200 station lines. Requests for
	this service beyond 200 station lines may be provided on a Special Ser-
	vice Arrangement Request basis, where facilities permit.

- (2) Call Transfer Individual, Consultation Hold and Add-On will function only on an incoming call from the telecommunications network.
- (3) Add-On provides for a maximum of one additional station line being added at one time to an existing incoming call.
- (4) The minimum monthly charge is for 20 station lines per location.
- (5) Station Hunting is limited to not more than ten station lines, with hunting arrangement. Requests for service above ten consecutive station lines may be provided on a Special Service Arrangement Request basis.
- (6) Refer to Paragraph 31.9.1, T., for obsolescence regulation.
- (7) Available only when switching equipment other than that serving the primary location is provided at the secondary location.
- (8) Refer to the Local Exchange Tariff for applicable rates for PBX trunks.
- (9) Refer to Section 24 of this Tariff for applicable charges for Touch-Tone-Service.

<u> </u>		<u></u>			
Issued:	NOV 3 0 1984	Effective:	DEC 31	1984	DEC 3 1 1934
	Southweste	N, President-Missou rn Bell Telephone (t. Louis, Missouri		11	lic Service Commission

General Exchange Tariff Section 31 Original Sheet 32 In EUSE

OBSOLETE SWITCHING SYSTEMS

(CP) 31.9 COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS-(Continued) NOV 3 0 1824

31.9.6 Optional Features(1)

MISSOURI

The following Optional Service Features are available Child Schuler Sc

		Monthly Rate	Complex Installation/ Move Charge	Service and Equipment Charge
(NR)	- Additional Attendants Position (RXXWC) or (RXYWC)(4) (Console or			
	(Switchboard)	\$383.90		\$4.75
(NR)	- Attendant Conference			
	Arrangement (RKT)(3)	38.40	\$125.00	4.75
(NR)	- Busy Indication Lamps (per group of 20 lamps,			
(NR)	per position) (PQ5)(2). - Cable Turning Section,	7.60		4.75
	each (CQM) (5) (6)	8.60	50.00	4.75

GANCELLED

JUL 1 1986 BY 2 200 R S. 27 PUBLIC SERVICE COMMISSION OF MISSOURI

- (1) Refer to Paragraph 31.9.1, T., for obsolescence regulation.
- (2) A Complex Installation/Move Charge of \$200.00, per installation, will apply, regardless of the number of lamps provided, when installed subsequent to the initial installation and when no other switchboard or dial equipment additions are being made at the same time. The Busy Indication Lamp can be installed on switchboards only.
- (3) This is an attendant-controlled conference arrangement which is limited to not more than one conferee's telephone involved in the conference call being located on premises which are noncontiguous to the attendant's location.
- (4) A three-year termination contract is applicable in the amount of \$5,550.00.
- (5) Applies only to new installations of switchboard positions.
- (6) Not applicable to switchboards that are multiplied.

Issued:	NOV 3 0 1984	Effective: DEC 31 19	4 FILED
	Southweste	DN, President-Missouri Division ern Bell Telephone Company	DEC 5 1 1824
	۲	St. Louis, Missouri	Public Service Commission

General Exchange Tariff Section 31 Original Sheet 33

REGEIVED

(CP) 31.9 COMMUNICATIONS SERVICE PACKAGE PREMISES SYSTEMS-(Continued)

31.9.6 Optional Features(1)-(Continued)

		Monthly	Comple Public S Installation/	MISSOURI ervice Cutifinission
		_Rate	Move Charge	Charge
(NR)	- Conference Calling (EAN)(2)	\$30.75	\$150.00	\$4.75
(NR)	- Code Call Access Arrangement, each			
(MT)	<pre>(CCX)(3) - Custom Calling Service Features Speed Calling Attendant Position(s) and first 25 unre- stricted main sta- tion lines equipped</pre>	86.40	300.00	4.75
	(E2G)(4)	46.80 GANCE	50.00	4.75

JUL 1. 1986 Br gans R.S. 2

- (1) Refer to Paragraph 31.9.1, T., for obsofestence regulation.
- (2) This is a station-controlled conferencing arrangement. This conferencing equipment provides for the simultaneous connection of several Communications Service Package stations on the same premises with the switching equipment. Connection of lines other than as stated above is subject to the limitations of the equipment, and where such equipment can be connected to Telephone Company Exchange Service, Private Line Service or Long Distance Message Telecommunications Service, the responsibility of the Telephone Company shall be limited to the furnishing of facilities suitable for Exchange Service, Private Line Service in a manner proper for such services.
- (3) Regulations covering the provisions of this service are applicable as provided in Section 4 of this Tariff.
- (4) A Service Charge of \$10.00 applies for each group of ten Speed Calling Numbers or fraction thereof that are simultaneously added, removed or changed subsequent to the initial installation of the Speed Calling feature.

	N4	
Issued: NOV 30 1984	Effective: DEC 3 1 198	A 序目上注向
	ON, President-Missouri Divisio ern Bell Telephone Company	n DEC C 1 1934
	St. Louis, Missouri	Public Service Commission



	r.	S.C. MONO. 35		
Ф	No supplement to this tariff will be issued except for the purpose of canceling this tariff.			nge Tariff Section 31 1 Sheet 34
•		E SWITCHING SYSTEMS		EUMEM
(CP)	31.9 COMMUNICATIONS SERVICE PACE	AGE PREMISES SYSTEM	S-(Continued)	
	31.9.6 Optional Features(5)-(0	Continued)	NOV 3	³ 0 1934
		Monthly Rate	Complex Mis Instablacion/ Move Charge	SOURI _{Equipment} COMMISSINTE
(MT)	- Custom Calling Ser- vice Features- (Continued) Speed Calling- (Continued) For each addi- tional main station line			
(MT)	equipped(2) Call Hold and Call Pickup, per main station line equipped	. \$1.95	\$1.00(1)	\$4.75
(MT)	(EA2)(3) Call Forwarding - Variable, per main station line	. 1.60	1.00(4)	3.75
	equipped (EAT)	ANGELLED	1.00(4)	3.75
	BY (1) Establishes the initial repep (2) A Service Charge of \$10.00 ap	ITELED SERVICE COMMISSION	 I	**
	 (2) A Service Charge of \$10.00 ap Numbers or fraction thereof t changed subsequent to the ini (3) These features are provided a a station line must be equipp for Call Pickup. All station equipped with the same featur Charge with apply to each to 	hat are simultaneou tial installation o s a group per stati ed for Call Hold be lines in that same ces; therefore, the	sly added, remove f the Speed Calli on line equipped fore it can be ec group are automa above rate and In	ed or ing feature. ; however, quipped atically nstallation

ť

Charge will apply to each station line in the "Dial Pickup Group."

(4) When Call Hold, Call Pickup and Call Forwarding-Variable are installed simultaneously, a Complex Installation/Move Charge of \$1.00 per main station line equipped will apply in lieu of the above Complex Installation/Move Charges.

(5) Refer to Paragraph 31.9.1, T., for obsolescence regulation.

(S) NOV 30 1984Effective: DEC 3Issued: NOV 30 1984By R. D. BARRON, President-Missouri DivisionDEC 3 1 1984DEC 3 1 1984Southwestern Bell Telephone CompanySt. Louis, Missouri

	No supplement to this tariff will be issued except for the purpose of canceling this tariff. OBSOLETE SWI	TCHING SYSTEM	RE	ange Tariff Section 31 Mal Sheet 35
(CP)	31.9 COMMUNICATIONS SERVICE PACKAGE P	REMISES SYSTE	MS-(Continued)	IOV 3 0 1924
	31.9.6 Optional Features(1)-(Contin	ued) Monthly <u>Rate</u>	Comple Sublic Installation/ Move Charge	MISSOURI Service Commission Equipment Charge
(MT)	- Custom Calling Service Features-(Continued) Call Transfer - Individual - All Calls, Consultation Hold - All Calls and Three-Way Calling(3) Per 100 main station lines equipped or fraction thereof (E2E)(2)	\$46.80		\$4.75
(NR)	- Dictation Access Arrange- ment, each (RC8)(4)	38.40	\$200.00	

Ĩ

GANCELLED	
JUL 1 1986	
BY 2 MD R.S. 27 PUBLIC SERVICE COMMISSION OF MISSOURI	
 Refer to Paragraph 31.9.1, T., for obsolescence regulation. This feature is available to all station lines in a system as provided only on a system basis. Available only to Series 300 installations. Obsolete - applicable to existing installations at existing 1 	
for existing customers.	
Issued: NOV 30 1984 Effective: DEC 31 1984	DEC 3 1 1984
By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri	Public Service Commission

.

General Exchange Tariff Section 31 Original Sheet 36

	OBSOLETE SWITC	HING SYSTE	MS	REGE	IVED
(CP)	31.9 COMMUNICATIONS SERVICE PACKAGE PREM	ISES SYSTE	S-(Cont	inued)	
	31.9.6 - Optional Features(1)~(Continue	d)		NOV 3	J 1924
		Monthly Rate	Install	plex MISS Public Servic	SOUPLvice and e Commission Charge
(NR)	- Paging Access Arrange- ment, each (55N)(2)(5)	\$ 28.80	\$150.	.00	
(NR)	- Reserve Power(3) Workable Station Capacity	+	7 - 0 0 0	••••	
	20-499 (FLJ)	239.90			\$4.75
	Over 499 (FLJX1)	479.75			4.75
(NR)	- Station Direct Access to WATS, per line (WLA)(4)	15.35			4.75

- BELLED (1) Refer to Paragraph 31.9.1, T.,
- (2) Regulations covering the provisions of this service are applicable as provided in Section 4 of this Tariff, 1 1986 (3) A three-year termination contract is applicable as follows:
- ome S.JL

Workable Station Line Capacity	BY AMOUNTN
20 - 499	PUBLIC SERVICE COURS, 700.00
Over 499	13,350.00

This feature provides an alternate independent source of power to maintain telephone service (excluding certain attendant supervisory functions and key telephone arrangements) during a commercial power failure. The Telephone Company agrees to make every reasonable effort to assure continued operation until such time as commerical power is restored.

- (4) A Complex Installation/Move Charge of \$50.00 is applicable if installed subsequent to the initial installation of the packaging system.
- (5) Obsolete applicable to existing installations at existing locations for existing customers

CAID	ting customers.	F	PUL EM
Issued:	NOV 3 0 1984	Effective: DEC 31 1984	DEC 3 1 1984
		RCN, President-Missouri Division tern Bell Telephone Company St. Louis, Missouri	Public Service Commission

- -

	No supplement to this tariff will be issued except for the purpose of canceling this tariff.					Orig	change Tariff Section 31 <u>inal Sheet 37</u> 呂倪氏[[]]に同
				OBSOLET	E SWITCHING SYST		EGEIVED
(CP)	31.9 CO	MMUN I C	ATIONS SERVICE	PACKAGE PR	EMISES SYSTEMS-(Continued)	10V 5 0 1934
	31.9.7	Com	nunications Ser	vice Packag	e Premises Syste		Service Commission
					e Systems will b and conditions s		
	Α.	Commu	nications Servi	.ce Package	Systems(1)		
		1. Pı	imary Location	1			
			imary location ne contract sch	-	ire a terminatio	on contract	based on
		2. Se	condary Locati	lon			
		50			equire a separate station line cour		
(NR)		3. Co	ontract Schedul	_e(2)			
			Workable Sta Forecas	ation Line (st at One Ye			on Charges 0, 200, 300
				20-27		\$ 19,80	
				28-41 42-55			50.00 50.00
				56-72		•	0.00
				73-90		42,95	50.00
				91-108		m 49,40	00.00
					ANGELLE	<u>وں</u> 62,50	00.00 00.00
				145-180	~	109,00	
				271-360	JUL 1 1986	141,80	
				361-450	and a g	1 175,0	00.00
				451-540 ₈₁	JECS. J	207,90	
				541-630 _{PU} 631-720	BLIC SERVICE COA MID-	× 240,8 NO. 273,8	
				721-810		307,0	
				811-900		339,7	
				901-1000		372,9	00.00
	(1) Cont	tract	term is three	vears.			
	(2) For	each	additional gro	up of stati	on lines with a		
	capa	acity	or fraction th	ereof excee	ding the contrac		
_			on contract am			aulattes	
	(3) Kefe	er to	raragraph 31.9	.1, 1., ior	obsolescence re	guiacion.	
-	Issued:	NO	V 30 1984		Effective;DEC	3 1 1984	DEC 3 1 1934
							L.

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff. General Exchange Tariff Section 31 Original Sheet 38

4.75

OBSOLETE SWITCHING SYSTEMS

(00)	<u></u>	101170			
(CP)	31.9 COM	FIUNICA	TIONS SERVICE PACKAGE PREMISES SYSTEMS	REGENTED	
	31.9.8	Facil	ity Terminations		
	A. T	lie Lir	les	NOV 30 1984	
	1	pro	e lines are furnished subject to the av wided no unusual expense is involved.	_ MISSOURI	
	2	sy's pre	e lines may terminate in Communications tems of the same or different customer mises. For private line charges appli tion 2 of the Private Line Service Tap	r and on the same or different icable to tie lines, see	
	3		e following regulations apply in connected associated with Communications Serv		
		a,	Communications Service Package System PBX or Centrex System are arranged an vice Package end, for standard operation connected with trunks, off-premises s	t the Communications Ser- tion and transmission when	s
)		Ъ.	Tie lines connecting Communications a Communications Service Package System System of different customers, are fu that such tie lines are provided only volved is connected to the central of	m with another PBX or Centrex urnished with the understanding y if each of the systems in-	2
(NR)	4	4. Uni	versal Tie Line Terminal Charges		
			e following rates apply for a universa sociated with a Communications Service		
			GANCELL Rate	Complex Service and Enstallation/ Equipment Move Charge Charge	1
		Wi of:	th operator and/or central JUL 1 1986 fice access, each (TLL) \dots 9 \$77.00	\$53.25 \$4.75	

OF MISSOURI

(1) Charges equivalent to the charges for these universal tie line terminals will apply for the termination of station lines of a Communications Service Package System in attendant positions and/or switching equipment of another PBX or a Centrex System. The Telephone Company shall not be responsible for the through transmission of signals or for supervision when facilities are arranged in this manner.

office access, each (TLN) UBLIC SERVICE 02045AISLION 53.25

(2) Available where facilities permit.

Without operator or central 7

(3) Not intended for use in terminating interstate Enhanced Private Switched Communications Service (EPSCS) channels, Type A. Not available to customers served by No. 101 ESS.

served by No. 101 ESS.	FILED
Issued: NOV 30 1984 Effective: DEC 31 1984	DEC 3 1 1984
Der D. D. DADDON, Dessendent Misserveri Diesisis	Public Sarvice Commission

ł

tar exc	iff wil ept for	ent to this l be issued the purpose ng this tariff.	General Exchange Tariff Section 31 Original Sheet 39
		OBSOLETE SWITCHING SY	REGEIVED
(CP) 31.	9 COMM	UNICATIONS SERVICE PACKAGE PREMISES SY	STEMS-(Continued) NOV 3 3 1984
3	1.9.8	Facility Terminations-(Continued)	MISSOURI Public Service Commission
	B. Ne	twork Access Terminals	
	1.	Network Access Terminals are furnish Package Premises Systems to provide customer's SCAN- and CCSA-type netwo	station users access to the
	2.	Dial Network Access Terminals includ features(2):	le the following service
		a. Network Inward Dialing of Calls Service Package Premises System.	to station lines on the Communications
		b. Direct Outward Dialing of networ Communications Service Package S the network.	
		-	network calls from one station of e System to another station of that
		d. Discrete identification to the a	attendant of incoming network calls.
	3.	Access lines connecting attendant po or CCSA-type network are furnished f attendant positions and stations on	
(NR)	4.	Rates	
		The following rates and charges appl associated with a Communications Ser CCSA-type network:	
		GANG	thly Installation/ Equipment te 10 Give Charge Charge
		Communications Service Package Systems Dial network access terminals, each (901)(1) . PUBLIC. SF562	$\frac{R \cdot S \cdot \partial 7}{2^{CE} COMMISSION}$
		Manual termination in attendant	
		position only, per network access terminal (9CS) 32	2.40 65.00 4.75
		des incoming selector and rotary out-tr	
(2)) Provid	ed only where appropriate equipment is	
Iss	sued:	NOV 30 1984 Effectiv	ve: DEC 3 1 1984 DEC 3 1 1984
		By R. D. BARRON, President-Mis Southwestern Bell Telephor St. Louis, Misson	ne Company PUDES Service Commission

Southwestern Bell Telephone Company St. Louis, Missouri
-- · ·

- ---

.__ . .

(CP) (NR)	tariff wi except fo of cancel 31.9 COM 31.9.8 B. N	Facility Termir	OBSOLETE SWITCHING SY VICE PACKAGE PREMISES SY nations-(Continued) erminals-(Continued)		original REGEIV	Section 31 Sheet 40
		-	g rates apply for a netw nications Service Packag			
		With mult pre-empti Communica Package	work terminals: ti-level precedence ion calling: ations Service e Systems, each	\$90.20	\$150.00	\$4.75
		pre-empti Communi Package	multi-level precedence ion calling: ications Service e Systems, each	83.00	150.00	4.75
		Access only in with au supervi pre~emp	etwork access terminals lines will terminate an attendant position atomatic network ision, and/or precedence ption capabilities, each	36.10	50.00	4.75
		switched level pro calling:	JUL 1 BY 2 200	19.40 1986 <u>R.S.J</u> E COMMISSIC	25.00	4.75
			Ur «		FILL	ED
	Issued:		D. BARRON, President-Mis uthwestern Bell Telephon	e Company	UEU JI	
			St. Louis, Missou	ri	Funic contro	VVIIIIII VVIIIII

COMPETITIVE EXCHANGES

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

(AT)

General Exchange Tariff Section 32 2nd Revised Sheet 1 Replacing 1st Revised Sheet 1

32.1	BUSINESS EXCHANGES			
	The following exchanges ar	e classified as competitive for business services:		
(AT)	Antonia Bonne Terre Camdenton Cape Girardeau Chesterfield Clever Eldon Eureka Farmington Fenton Festus-Crystal City Flat River Fredericktown Fulton Grain Valley Gravois Mills Greenwood Harvester Herculaneum-Pevely High Ridge Imperial Jackson	Joplin Kansas City Lake Ozark-Osage Beach Manchester Marble Hill Maxville Monett Nevada Pacific Perryville Pond Poplar Bluff Scott City Sedalia Sikeston Smithville Springfield St Charles St Genevieve St Joseph St Louis Valley Park Washington		
(RT) (RT)				

Issued: September 28, 2005

Effective: October 10, 2005

Filed

MO PSC

By CINDY BRINKLEY, President-SBC Missouri Southwestern Bell Telephone, L.P., d/b/a SBC Missouri St. Louis, Missouri

CANCELLED June 29, 2007 TO-2002-185 Missouri Public Service Commission

TO-2006-0093

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 32 1st Revised Sheet 1 Replacing Original Sheet 1 Original Sheet 13 and 1st Revised Sheet 14 2nd Revised Sheet 15 Original Sheet 16 through Original Sheet 29 1st Revised Sheet 30

RADIO COMMON CARRIERS

The Radio Common Carrier, Section 32 of the General Exchange Tariff, P.S.C. Mo.-No. 35, has been withdrawn, in its entirety, and the contents completely removed. This section has been replaced, by P.S.C. Mo.-No. 40, Radio Common Carrier Interconnection Tariff.

CANCELLED

October 10, 2005

MISSOURI PUBLIC SERVICE COMMISSION

Issued: November 27, 1990

By R. D. Barron, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

RADIO COMMON CARRIERS

General) Exchange (Tariff

MISSUURI

CANCELLED^{Original Sheet 1}

DEC 1 1990

PR5

- ----

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

32.1 REGULATIONS

32.1.1 Definitions

Definitions <u>Actual Cost</u> - Denotes all costs charged again SS Specific case of <u>Special construction</u> including special construction, including any appropriate taxes.

Annual Underutilization Liability - Denotes a per-unit amount which may be billed annually if fewer services are in use utilizing specially constructed facilities at filed tariff rates than were originally specially constructed.

Authorized Services - Those domestic public land mobile radio communications services which the carrier may now or hereafter lawfully provide on an interconnected basis.

Carrier's System - The communications system of the carrier used to furnish public land mobile radio services.

Class 5 Central Office - The central office trunking entity where telephone loops are terminated for purposes of interconnection to each other and to the network.

Company's System - The communications network of the Company and its connecting companies.

Connecting Circuit - The facility and associated service arrangement used to connect the carrier's system with the Company's system for the purpose of interchange local and intraLATA traffic.

Dial Access Line - This is equivalent to a regular business exchange service line between the Company's serving (Class 5) central office in . the exchange of interconnection and the carrier's control point. The exchange of interconnection is normally that exchange within which the control point of the carrier's land radio station is located.

Direct Inward Dial (DID) - An operational feature that provides direct dial access to mobile radio and paging terminals by transmitting address signaling from the serving C.O. to the terminal. This is generally referred to as "outpulsing."

JAN = 1 (G3)Estimated Cost - Denotes all estimated costs that will be incurred in inc18d3ns 215 3 providing a specific case of special construction Public Service Commission appropriate taxes.

Facilities - Denotes any cable, poles, conduit, microwave or carrier equipment, wire center distribution frames, central office switching equipment, etc., utilized to provide services offered under the tariffs referenced by this Tariff.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

COMPETITIVE EXCHANGES

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 32 1st Revised Sheet 2 **Replacing Original Sheet 2**

32.1 **BUSINESS EXCHANGES** The following exchanges are classified as competitive for business services: Archie Ash Grove Billings Boonville

Carthage Cedar Hill Chaffee Chillicothe De Soto Dexter

Farley

Hillsboro

Excelsior Springs

Gray Summit Hannibal

(AT)

CANCELLED June 29, 2007 TO-2002-185

Missouri Public Service Commission

(AT)

Issued: August 30, 2005

Effective: October 29, 2005

By CINDY BRINKLEY, President-SBC Missouri Southwestern Bell Telephone, L.P., d/b/a SBC Missouri St. Louis, Missouri



Kennett
Kirksville
Linn
Marionville
Marshall
Mexico
Moberly
Montgomery City
Neosho
Portage Des Sioux
Richmond
St Clair
Union
Ware
Webb City

TO-2006-0102

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General-Exchange-Tariff-CANCELLED Griginal Sheet 2 DEC 1 1990 DEC 2 9 1883 RADIO COMMON CARRIERS INTRS# BY Public Service Commission OURI MISSO BELIC Service Commission

32.1 REGULATIONS-(Continued)

32.1.1 Definitions-(Continued)

<u>Installed Cost</u> - Denotes the total investment (estimated or actual) required by the Telephone Company to provide specially constructed facilities.

<u>Maximum Termination Liability</u> - Denotes the maximum amount which may be billed if all services using specially constructed facilities are terminated prior to the expiration of the Maximum Termination Liability Period.

Maximum Termination Liability Period - Denotes the length of time for which a Termination Charge may apply if all services using specially constructed facilities are terminated.

<u>Net Salvage</u> - Denotes the estimated scrap, sale or trade-in value, less the estimated cost of removal. Cost of removal includes the costs of demolishing, tearing down or otherwise disposing of the material and any other applicable costs. Since the cost of removal may exceed salvage value, net salvage may be negative.

Nonrecoverable Cost - Denotes the cost of specially constructed facilities for which the Telephone Company has no foreseeable use should the service be terminated.

Normal Cost - Denotes the estimated cost to provide services using normal construction.

<u>Normal Construction</u> - Denotes all facilities the Telephone Company would normally use to provide service in the absence of a request for special construction.

Permanent Facilities - Denotes facilities providing service for one month or more.

Point of Connection - The point of connection between the connecting circuits provided by the Company and the facilities of the carrier.

Recoverable Cost - Denotes the cost of the specially constructed facilities for which the Telephone Company has a forseeable reuse either in-place or elsewhere, should the service be terminated. 83-253

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

Public Service Commission

(AT)

(AT)

General Exchange Tariff Section 32 1st Revised Sheet 3 Replacing Original Sheet 3

COMPETITIVE EXCHANGES

32.2 RESIDENTIAL EXCHANGES

The following exchanges are classified as competitive for residential services:

Advance	Pacific
Bell City	Perryville
Chesterfield	Pocahontas-New Wells
Delta	Pond
Eureka	Smithville
Farmington	Springfield
Fenton	St Charles
Fredericktown	St Genevieve
Harvester	St Joseph
Kansas City	St Louis
Manchester	Valley Park
Monett	Washington
Nevada	Wyatt

Issued: September 28, 2005

Effective: October 10, 2005

By CINDY BRINKLEY, President-SBC Missouri Southwestern Bell Telephone, L.P., d/b/a SBC Missouri St. Louis, Missouri



TO-2006-0093

General Exchange Farif Original Sheet CANCELLED DEC 29 1983 RADIO COMMON CARRIERS 1990 DEC 1 [AISSOURI BY LAT Public Service Commission

32.1 REGULATIONS-(Continued)

32.1.1 Definitions-(Continued)

MISSOURI Reliable Service Area - The reliable service area of the carrier's base station as described in Section 22.504 of the Federal Communications Commission's Rules.

Termination Charge - Denotes the portion of the Maximum Termination Liability that is applied as a nonrecurring charge when all services are discontinued prior to the expiration of the specified liability period.

32.1.2 Interchange of Traffic

At the carrier's request, the Company and carrier will physically connect their facilities and interchange traffic originating or terminating on carrier's system in connection with carrier's authorized services. Such interconnection shall be in accordance with the service, operating and facility arrangements set forth hereinafter.

Interchanged local and intraLATA traffic shall be handled only over connecting circuits provided, except in the case of occasional emergency traffic of carrier's subscribers.

Where the state franchised area or state authorization of carrier differs or is modified after the effective date of this Tariff so as to differ from its reliable service area, the terms and conditions of interconnection may be modified to recognize the extent of such franchise or authorization.

32.1.3 Facilities and Arrangements

A. Subject to the availability of facilities and the reasonable requirements of the Company for its telecommunications services, Company will provide to carrier, upon request, those facilities and arrangements described herein which are necessary to establish the physical connection and interchange of local and intraLATA traffic provided for herein and other intraLATA facilities carrier may require for operation of its system such as remote transmitter and receiver links. The facilities and arrangement, the location(s) of such connection(s), including maintenance and operating points of demarcation and the additional facilities provided for use in carrier's system are set forth in JAN - 1 1934 Paragraphs 32.1.20, A. and B. 83-253

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Public Service Commission

General Exchange Tariff Section 32 1st Revised Sheet 4 Replacing Original Sheet 4

CANCELLED June 29, 2007 TO-2002-185

Missouri Public Service Commission Issued: August 30, 2005

COMPETITIVE EXCHANGES

32.2 RESIDENTIAL EXCHANGES

The following exchanges are classified as competitive for residential services:

Antonia	Herculaneum-Pevely
Archie	High Ridge
Ash Grove	Hillsboro
Billings	Imperial
Bonne Terre	Jackson
Boonville	Joplin
Camdenton	Kennett
Cape Girardeau	Kirksville
Carthage	Lake Ozark-Osage Beach
Cedar Hill	Marionville
Chillicothe	Marshall
Clever	Maxville
De Soto	Mexico
Dexter	Moberly
Eldon	Neosho
Excelsior Springs	Poplar Bluff
Farley	Portage Des Sioux
Festus-Crystal City	Richmond
Flat River	San Antonio
Fulton	Sedalia
Grain Valley	Sikeston
Gravois Mill	St Clair
Gray Summit	Union
Greenwood	Walnut Grove
Hannibal	Ware
	Webb City

(A'T)

(AT)

Effective: October 29, 2005

By CINDY BRINKLEY, President-SBC Missouri Southwestern Bell Telephone, L.P., d/b/a SBC Missouri St. Louis, Missouri



TO-2006-0102

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

32.1.3 Facilities and Arrangements-(Continued)Public Service Commission

- B. In interchanging two-way mobile traffic, Company will extend and connect its facilities with those of carrier between:
 - 1. A Class 5 central office of any telephone exchange within a LATA whose rate center is located in carrier's reliable service area, and
 - 2. The message center(s) or control point(s) in or serving carrier's reliable service area in the same LATA.
- C. In interchanging one-way signaling traffic, Company will extend and connect its facilities with those of carrier between:
 - 1. A Class 5 central office of any telephone exchange within a LATA which a paging device is serviced by carrier's integrated single or multitransmitter extended area paging system, and
 - 2. The control point(s) in or serving carrier's extended area system in the same LATA.
- D. Any facility or arrangement provided hereunder shall be of the type normally utilized in the furnishing of the Company's intraLATA telecommunications services including, but not limited to: effective twowire and four-wire voice grade and voice grade data facilities, wire pair facilities to the extent offered under tariff to other carrier, central office outpulsing facilities and exchange access lines or trunks from the central office serving carrier to carrier's premises.
- E. Central office facilities and blocks of telephone numbers provided by Company for carrier's use shall be assigned by Company and may be changed to meet the reasonable operating and service requirements of Company. Carrier shall not acquire any proprietary interest in numbers assigned for its use. Where such changes are to be made, Company shall advise carrier in writing not less than 60 days in advance and will coordinate the changes with carrier.
- F. Central office codes used with telephone numbers furnished to carrier may be used by Company in providing its subscriber services and for other purposes.
 JAN - 1 '934

83 - 253 Public Service Commission

General Exchange Tariff

CANCELBED EIVED

DEC 1 1990 DEC 2 9 1033

BY PA

Section 32

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

General Exchange Tariff

011210315

DEC 1 1990

Section-32

DEC 2 9 1983

L⁴ISSOURI

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS CANCELLED

32.1 REGULATIONS-(Continued)

BY LOURS Public Service Commission 32.1.3 Facilities and Arrangements-(Continued)

- G. Carrier shall be provided telephone numbers in MissouAl and incremental blocks of 100 numbers each, but Company and carrier may agree to the provision of numbers in greater quantities. Carrier shall furnish Company its number requirements for planning purposes, but shall not be required to pay for numbers in advance of their assignment by Company in order to guarantee number availability. Within the number blocks assigned under Paragraph E., preceding, the assignment and use of such numbers of carrier to its subscribers for its authorized services shall be the responsibility of carrier subject to coordination with Company for efficient operation with Company's system.
- H. Company will cooperate with carrier in the development of any necessary specialized telecommunications facilities or arrangements reasonably required for the physical connection and interchange of traffic provided hereunder.

32.1.4 Use of Facilities and Services

- The connecting circuits provided hereunder shall be used only for the Α. handling of interchanged local and/or intraLATA traffic originating or terminating on carrier's system in connection with carrier's authorized services. Such circuits may, however, be used occasionally or incidentally for incoming calls concerning administrative matters related to carrier's authorized services.
- B. Connecting circuits, facilities or arrangements provided shall not be used, switched or otherwise connected together by the carrier for the provision of through calling from a landline telephone to another landline telephone, nor shall they be switched or otherwise connected together by the carrier for the provision of through calling from a landline or mobile unit located in one reliable service area to a landline telephone or mobile unit in another reliable service area, except as specifically provided for in this Tariff.
- C. Connecting circuits, facilities and arrangements provided carrier-by-Company shall not be used knowingly for any purpose or in any manner, directly or indirectly, in violation of law or in aid of any unlawful act or undertaking. JAN - 1 1934

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

83 - 253 Public Service Commission

General Exchange Tay Iff Section 32 Original Sheet 6 DEC 29 1983 CANCELLED MISSOURI DEC 1 390 lic Service Commission

32.1 REGULATIONS-(Continued)

32.1.5 Charges for Facilities and ArrangementsPublic Service Commission MISSOURI

RADIO COMMON CARRIERS

- A. The facilities and arrangements provided by Company to carrier and all charges, therefore, are set forth hereinafter in Section 32.3. Carrier agrees to pay the Company all charges specified in Section 32.3 within 30 days of the date of the statement rendered for those charges.
- B. All monthly charges provided for in Section 32.3 shall be billed by Company monthly, in advance, except that charges due for the initial month, or a portion of the initial month, during which new items are provided will be included in the next bill rendered.
- C. In the event that the facility or arrangement has been in service for less than one month, the full monthly charge shall be due on termination, together with any applicable nonrecurring charges. In the event prior to initiating service to the public, carrier cancels an order for a facility or arrangement prior to placing it in service; and Company has incurred costs in connection with the facilities or arrangements to be provided, carrier shall reimburse Company the full charges for one month, together with any applicable nonrecurring charges or the reasonable costs, less net salvage, actually incurred by the Company if such costs are greater than the monthly and applicable nonrecurring charges.
- D. For the purposes of Paragraph C., preceding, the term "costs" shall include the nonrecoverable cost of equipment and material ordered, plus the nonrecoverable cost of installation and removal including the cost of engineering, labor, supervision, transportation, rights-of-way and other associated costs.

32.1.6 Charges, Billing and Payments for Interchanged Calls

- A. The charges to subscribers of Company and of carrier for an interchanged call shall be determined as set forth in this paragraph and shall be set forth in their respective tariffs applicable to the type of traffic interchanged.
 - 1. The total charge for an interchanged call shall be the Company's charge for its portion of the service plus the carrier's charge for its portion of the service. Except as provided in 2., Hellow ing, neither Company nor carrier may impose charges (other than those for facilities and arrangements as set forth in Section 1323) upon the other carrier or upon subscribers of the other carrier in connection with interchanged calls provided hereunder. Service Commission .

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

General Exchange Tariff 而尼向Section-32 DEDEDC 7990 BY 1240 C Public Service Com

32.1.6 Charges, Billing and Payments for Interchanged Calls-(Concerned)

- A. (Continued)
 - 2. The charges for intraLATA message toll telephone service shall be the charges set forth in Company's applicable message toll tariff between the point of origination or termination on Company's system and the point or points of interchange with carrier's system, plus carrier's charge for use of its system during such message.
 - a. For messages billable to stations on Company's system or to Company's subscribers (i.e., sent-paid from Company stations or placed on a received-collect or Company credit card basis to Company stations), Company shall bill and collect from its subscriber its charges for its portion of the service.
 - b. For messages sent-paid from stations on carrier's system or placed on a received-collect basis to such stations, Company shall bill carrier for Company's charges for its portion of the service, and carrier will pay such charges without allowance for uncollectibles. Carrier shall bill and collect its charges for its portion of the service and Company's charges billed to carrier.
 - c. Where carrier bills its subscriber for intraLATA message toll telephone service, carrier shall bill at the Tariff rates of the Company without discount or surcharge.
- B. Necessary timing and ticketing of Company's portion of an interchanged call shall be performed by Company. Necessary timing and ticketing of carrier's portion of a through call shall be performed by carrier.
- C. An itemized statement of charges payable to Company by carrier for Company's portion of through calls originating sent-paid from stations on carrier's system or placed on a received-collect basis to stations on carrier's system shall be rendered monthly to carrier by Company. Carrier shall remit the full amount of all stated charges to Company within 30 days following the date of the statement.



Issued:

DEC 29 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

32.1.7 Testing

public Service Commission BY Public Service Commission MISSOURI

DE0 1

CANCELLED

1990MISSOURI

hange Tariff

Public Service Commission

Section 32

The Company and the carrier each may make reasonable tests and inspections of its facilities and may, upon notice to and coordination with the other, temporarily interrupt the facilities being tested or inspected. When cooperative testing is requested by either party, such testing shall be done in accordance with Paragraph 32.1.18.

32.1.8 Trouble Reporting

Subscribers of the carrier shall be instructed to report all cases of trouble to the carrier. The carrier shall handle trouble reporting and advise the Company in accordance with the procedures established in Paragraph 32.1.18.

32.1.9 Equipment Space and Power

The carrier shall furnish or arrange to have furnished to the Company, at no charge, equipment space and electrical power required by the Company to provide facilties. The selection of ac or dc power shall be mutually agreed to by the carrier and the Company. The carrier shall also make necessary arrangements in order that the Company and its agents will have access to such equipment space at reasonable times for installing, inspecting, repairing or removing its facilities.

32.1.10 Liability and Indemnity

- Neither party assumes any liability for any act or omission of the other Α. in the furnishing of its service to its subscribers.
- The parties' performance shall be excused by labor difficulties, govern-Β. mental orders, civil commotions, acts of God and other circumstances beyond the parties' reasonable control, subject to the interruption allowance provisions of Paragraph 32.1.12, following.
- C. The liability of the Company for damages arising out of delays in installation, maintenance or restoration of facilities, services or arrangements or out of mistakes, omissions, interruptions, or errors or defects in transmission occurring in the course of providing such facilities, services or arrangements shall in no event exceed the amount of the allowance, if any, available under Paragraph 32.1.12, following. 83-253

Issued:

DEC 2 9 1983

JAN O 1 1921 Effective:

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

32.1.10 Liability and Indemnity-(Continued) Public Service Commission MISSOURI

- D. The Company shall reimburse the carrier for damages to premises or equipment of the carrier resulting from the provision of facilities, services or arrangements by the Company on such premises or resulting from the installation or removal thereof if caused by the negligence or willful act of the Company.
- E. The carrier shall reimburse the Company for damages to facilities of the Company if caused by the negligence or willful act of the carrier due to malfunction of any facilities or equipment provided by other than the Company. The Company will, upon reimbursement for damages, cooperate with the carrier in prosecuting a claim against the person causing such damage, and the carrier shall be subrogated to the right of recovery by the Company for the damages to the extent of such payment.
- F. The carrier shall reimburse the Company for any loss through theft of facilities provided on the carrier's premises.
- G. The parties shall cooperate with each other in the defense of any suit, claim or demand by third persons against either or both of them arising out of the connection arrangements and interchange of traffic hereunder including, without limitation, Workmen's Compensation claims, actions for infringement of copyright and/or unauthorized use of program material, libel and slander actions based on the content of communications.
- H. Neither party shall be required to reimburse the other for any claim or loss pursuant to this paragraph where the amount in controversy is less than \$100.

32.1.11 Patents

A. With respect to claims of patent infringement made by third persons, carrier shall defend, indemnify, protect and save harmless Company from and against all claims arising out of the combining with or use in connection with, the facilities, services or arrangements of any circuit, apparatus, system or method provided by carrier or its subscribers.



EGEIVED

original Sheet 9

[.]:SSOURI

Public Service Commission

CANCEL

DEC 1

Section 32

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

32.1 REGULATIONS-(Continued)

DERENSection 32 Deriginal Sheet 10 RADIO COMMON CARRIERS DEC 1 1990 MISSOURI BY PACS Service Commission Public Service Commission 32.1.11 Patents-(Continued)

__General_Exchange_Tariff

B. No license under patents is granted by Company to carrier or shall be implied or arise by estoppel in carrier's favor with respect to any circuit, apparatus, system or method used by carrier in connection with any facilities, services or arrangements. With respect to claims of patent infringement made by third persons, Company will defend, indemnify, protect and save harmless carrier from and against all claims arising out of the use by carrier of facilities, services or arrangements furnished by Company.

32.1.12 Allowance for Interruptions

When use of the facilities furnished by Company is interrupted due to trouble in such facilities, and such interruption is not caused by (a) the negligence of carrier or its subscriber or (b) the fault of facilities or equipment provided by carrier or its subscriber, carrier shall, upon request, be allowed a credit as follows:

- The amount of credit to carrier shall be an amount equal to the pro rata Α. monthly charge specified in Section 32.3 for the period during which the facility affected by the interruption is out of service.
- B. All credit for interruption shall begin from the time of actual notice by carrier to Company, in accordance with Paragraph 32.1.8, preceding, that an interruption of use has occurred. No credit shall be allowed for an amount of less than \$5.00.
- C. A credit shall not be applicable for any period during which carrier fails to afford access to the facilities furnished by Company for the purpose of investigating and clearing troubles.
- 32.1.13 Technical Specifications

Subject to any special arrangements pursuant to Paragraph 32.1.3, H., preceding, the design, installation, operation and maintenance of all circuits, equipment and other facilities of carrier and Company which are used in handling interchanged traffic shall be made in accordance with Bell System Technical Reference Number 43303, Fas from time to time may be amended, a copy of which has been provided to carrier by Company.

> JAN - 1 1934 83-253 Public Service Commission

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

32.1 REGULATIONS-(Continued)

32.1.14 Protection

Public Service Commission A. The characteristics and methods of operation of any circuits, facilities

BY

RADIO COMMON CARRIERS CANCELLED DEC 20 1003

DEC 1

1990

General Exchange Tariff DEDESECTION 32 Original Sheet 11,

VIISPOARI

Service Commission

- or equipment of the carrier connected with the services, facilities or equipment of the Company shall not interfere with or impair service over any facilities of the Company, its affiliated companies or its connecting and concurring carriers involved in its services, cause damage to their plant, impair the privacy of any communications carried over their facilities or create hazards to the employees of any of them or the public.
- If such characteristics or methods of operation are not in accordance Β. with A., preceding, the Company will, where practicable, notify the carrier that temporary discontinuance of the use of any circuits, facility or equipment may be required; however, when prior notice is not practicable, nothing contained herein shall be deemed to preclude the Company from temporarily discontinuing forthwith the use of a circuit, facility or equipment if such action is reasonable under the circumstances. In case of such temporary discontinuance, the carrier will be promptly notified and afforded the opportunity to correct the condition which gave rise to the temporary discontinuance. During such period of temporary discontinuance, allowance for interruption of service as set forth in Paragraph 32.1.12, preceding, is not applicable.
- C. The physical connection of facilities and interchange of traffic hereunder may be temporarily discontinued by either party upon 30 days' notice to the other for repeated or willful violation of or a refusal to comply with this paragraph or Paragraphs 32.1.2, 32.1.4 and 32.1.17.
- The party discontinuing will notify the Federal Communications Com-Ð. mission and the appropriate state regulatory body concurrently with the notice to other party of the prospective discontinuance for cause.

32.1.15 Records

Each party will keep adequate records of its operations and transactions and shall furnish to the other party such information as may be reasonably required for the administration of this flariffic-Carrier will, upon reasonable request, furnish copies or otherwise make available to Company its licenses and other federal and, if applicable, state regulatory authorization and its filed or to be 1934 filed tariffs or other published schedule of charges pertaining the traffic to be interchanged hereunder. Public Service Commissio

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

----General Exchange Tariff REGENERSECTION 32 Original Sheet 12

DEU 20 1000 CANCELLED

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

32.1.16 Deposits

- Public Service Commercial Commission A. The Company may, in order to safeguard its interests, require the SOURI carrier to make a deposit to be held by the Company as a guarantee of the payment of charges. Such deposit may not exceed an amount equal to the sum of (a) twice the amount of the monthly charges for facilities and arrangements specified in Section 32.3, plus (b) twice the estimated average monthly charges for local and toll messages billed to the carrier.
- B. The fact that a deposit has been made does not relieve the carrier from complying with the requirements for advance payments or the prompt payment of bills as specified elsewhere in this Tariff. At such time as the provision of facilities and arrangements to the carrier is terminated, the amount of the deposit will be credited to the carrier's account, and any credit balance which may remain will be refunded. At the option of the Company, such a deposit may be refunded or credited to the carrier at any time. In case of a cash deposit, simple interest at the annual rate of 9 percent shall be paid for the period during which such deposit is held by the Company.
- 32.1.17 Changes and Modifications
 - If the Company proposes to make any permanent changes in its operations Α. which would affect the carrier's operations or services once the facilities, arrangements, apparatus, equipment or any other item furnished by the Company to the carrier are installed, the Company shall give notice to the carrier in advance of any such changes advising when such changes will be made. All such changes shall be coordinated with the carrier. Where such changes are made for Company's purposes, and where practicable, temporary equipment or facilities shall be provided to the carrier by the Company at no additional charge. If the carrier proposes to make any changes in its operations or services which would affect Company's operations or services, the carrier shall give notice to the Company in advance of any such changes advising when such changes will be made. All such changes shall be coordinated with the Company.
 - B. Subject to the provisions of Paragraph A., preceding, carrier shallebe solely responsible, at its expense, for the overall design of its ser vices and for any redesigning or rearrangement of its services which may be required because of changes in facilities, operations of pro-1534 or maintenance characteristics of the facilities. Public Service Commission

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

32.1.18 Trouble Reporting, Installation and Testing ProceduresService Company

A. In order to facilitate trouble reporting and to coordinate the tepait to the commission of private line circuits (e.g., remote transmitter links) provided to Commission the carrier by the Company, the Company will designate a Plant Control SOURI Office (PCO) for such private line circuits.

General Exchange Tariff

Doriginal Sheet 13

TEA "TAIDE DE MINIM

Section 32

DEU 20 COCANCELLED

Public Service Commission

19**90**

- B. Where new private line circuits are installed, the PCO will ensure that continuity has been established and that appropriate transmission measurements have been made before advising the carrier that the new circuit is ready for service.
- C. In locations where requested by the carrier, the Company will furnish a direct seven-digit trouble reporting number for the designated PCO. This number will give the carrier access to the location where its facility records are normally located and where current status reports on any trouble reports are readily available. Alternative out-of-hours procedures will be established as required to ensure access by the carrier to a location which is covered and has the authority to initiate corrective action.
- D. To minimize outages on Company-provided exchange access and private lines, it is important that effective trouble reporting and clearing procedures be established. To that end:
 - When the carrier reports a trouble condition, it will first have used its best efforts to isolate the trouble to the Company's facilities. It will also advise the Company of the usage sensitivity of the circuit and the need for expedited clearance.
 - 2. In cases where the carrier indicates essential or important subscriber usage on lines provided by Company, the Company will attempt to expedite the clearance in a manner similar to its local procedures used to restore similar Company essential services.
 - 3. Company and carrier will make cooperative tests, as approFILED priate, to eliminate the necessity for either party to dispatch repair forces needlessly to distant unattended JAN - 1 1984 locations merely to isolate the trouble.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

P.S.C. Mo.-No. 35 CANCELLED

No supplement to this tariff will be issued except for the purpose of canceling this tariff. DEC 1 1990 DEC 1 Seneral Exchange Tariff BY Commission Section 32 Public Service Commission Street 14 MISS DEFINITION

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

JUN 2 7 1980

MISSUNRI

(CT) 32.1.19 Customer Owned Equipment Trouble Isolation Charges

- A. Carrier obtains from the Company intraLATA access fines Certific Companys in RCC equipment and intraLATA private line circuits connecting its control terminals to its transmitter/receiver isolation of each carrier's equipment and facilities for testing purposes. Both Company and carrier believe that because each carrier is a responsible, regulated communications common carrier, each should be responsible for isolating and clearing troubles on its own system.
- (CT) B. Carrier and Company recognize that isolation charges need not be applied if each carrier carries out its proper function and have, therefore, agreed to implement on a trial basis a procedure which will eliminate these charges.
 - C. Under this procedure, either Company or carrier upon discovering trouble in its service will respond to the trouble reported by isolating the problem to its own or the other carrier's system. Each carrier will clear the trouble in its own system prior to handing off the trouble to the other. However, if either carrier feels that the other is abusing the trouble reporting system and causing the other unreasonable or inordinate time and expense to find troubles which are ultimately determined to be in the reporting carrier's system, the aggrieved carrier may institute an isolation charge in accordance with the following procedure:
 - 1. Should a carrier believe that another carrier is not carrying out its responsibilities to isolate troubles to and clear troubles on its own system prior to reporting troubles to another carrier, the aggrieved carrier should notify the other carrier in writing that the accepted trouble reporting practices and procedures are being abused, with specific illustration of the abuse, and that he intends to assess isolation charges for any further abuses that occur.
 - 2. Upon receipt of the written notice by the other carrier, both carriers will meet as soon as possible to review the problem and take corrective action.
 - 3. If the carriers are unable to resolve the dispute, the aggrieved carrier will give written notice that he intends to implement 86 isolation charges by a specific date, but not less that ten days from the date of such notice.

Issued: JUN 27 1980

(TT)

(CT)

(CT)

Effective: JUL 1 1986

RADIO COMMON CARRIERS

DEU 2 LUU

Public Service for the service

.General Exchange Tariff

DECS Section 32

32.1 REGULATIONS-(Continued)

32.1.19 Maintenance of Service Charges

- A. Carrier obtains from the Company intraLATA access lines terminating in RCC equipment and intraLATA private line circuits connecting its control terminals to its transmitter/receiver isolation of each carrier's equipment and facilities for testing purposes. Both Company and carrier believe that because each carrier is a responsible, regulated communications common carrier, each should be responsible for isolating and clearing troubles on its own system.
- B. Carrier and Company recognize that maintenance charges need not be applied if each carrier carries out its proper function and have, therefore, agreed to implement on a trial basis a procedure which will eliminate these charges.
- C. Under this procedure, either Company or carrier upon discovering trouble in its service will respond to the trouble reported by isolating the problem to its own or the other carrier's system. Each carrier will clear the trouble in its own system prior to handing off the trouble to the other. However, if either carrier feels that the other is abusing the trouble reporting system and causing the other unreasonable or inordinate time and expense to find troubles which are ultimately determined to be in the reporting carrier's system, the aggrieved carrier may institute a maintenance charge in accordance with the following procedure:
 - 1. Should a carrier believe that another carrier is not carrying out its responsibilities to isolate troubles to and clear troubles on its own system prior to reporting, troubles to another carrier, the aggrieved carrier should notify the other carrier in writing that the accepted trouble reporting practices and procedures are being abused, with specific illustration of the abuse, and that he intends to assess maintenance charges for any further abuses that occur.
 - 2. Upon receipt of the written notice by the otherwarrier, both carriers will meet as soon as possible so the problem and take corrective action.
 - 3. If the carriers are unable to resolve the dispute, the apprieved carrier will give written notice that he intends to implement 253 maintenance charges by a specific date, but not less than ten days from the date of such notice.

Issued: DEC 29 1983

Effective: JAN 0 1 1984

General Exchange Tariff Section 32 2nd Revised Sheet 15 Replacing 1st Revised Sheet 15

RADIO COMMON CARRIERS

RECENCED

SEP 2 5 1989

DEC 1 1990

32.1 REGULATIONS-(Continued)

32.1.19 Customer Owned Equipment Trouble Isolation Charges-(Continued): Tublic Service Commission

C. (Continued)

(CT)

No supplement to this tariff will be issued

except for the purpose

of canceling this tariff.

- 4. After such date, a Customer Owned Equipment Trouble Isolation Charge, as found in the "Restoration of Service" section of the General Exchange Tariff, will be assessed for each incident where trouble reported to carrier or Company as being on its system, facilities or equipment is later found to be on the system of, facilities or equipment of the reporting carrier or Company NCELLED
- 32.1.20 Obligations of the Carrier

- Α.
 - 1. Carrier shall provide a voice intercept announcement or distinctive tone signals to the calling party when a call is directed to a number that is not assigned by carrier.
 - 2. Carrier shall provide a voice announcement or distinctive tone signals to the calling party when a call intended to activate a signal in a paging receiver has been received and accepted by carrier's terminal.
 - 3. When carrier's terminal is not able to complete calls because of a malfunction in the terminal or other equipment, carrier shall either divert the call to its operator or provide a recorded announcement to the calling party advising that the call cannot be completed.
 - 4. Carrier shall provide supervisory tones or voice announcements to the calling party on all calls, consistent with standard telephone industry practices.
- B. Line-Per-Terminal Systems
 - Carrier shall provide a voice intercept announcement to the calling 1. party when a call from Company's system is directed to a number that is unable to respond.
 - 2. Upon notification, Company will provide a reasonable period of intercept service when a call is directed to a number that has been disconnected from carrier's system.

Issued:

Effective: OCT 91 1989 07 1 1989 SEP 25 1990 By R. D. BARRON, President-Missouri Divisionublic Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

P.S.C. Mo,-No, 35

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

General Exchange Tariff Section 32 1st Revised Sheet 15 Replacing Original-Sheet 15

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

JUN 27 1980

WIZZONKI

Public Service Commission

32.1.19 Customer Owned Equipment Trouble Isolation Charges-(Continued) (CT)

C. (Continued)

(CT)

- 4. After such date, a Customer Owned Equipment Trouble Isolation Charge of \$25.00 will be assessed for each incident where trouble reported to carrier or Company as being on its system, facilities or equipment is later found to be on the system of, facilities or equipment of the reporting carrier or Company.
- 32.1.20 Obligations of the Carrier

Intercept and Acknowledgement of Calls by Carrier

- Outpulsing of Digits and End-to-End Signaling Arrangements Α.
 - 1. Carrier shall provide a voice intercept announcement or distinctive tone signals to the calling party when a call is directed to a number that is not assigned by carrier.
 - 2. Carrier shall provide a voice announcement or distinctive tone signals to the calling party when a call intended to activate a signal in a paging receiver has been received and accepted by carrier's terminal.
 - 3. When carrier's terminal is not able to complete calls because of a malfunction in the terminal or other equipment, carrier shall either divert the call to its operator or provide a recorded announcement to the calling party advising that the call cannot be completed.
 - EU Carrier shall provide supervisory tones or touce announcements to the calling party on all calls, consistent with standard telephone 4. industry practices. 5#13
- B. Line-Per-Terminal Systems
- Line-Per-Terminal Systems 1. Carrier shall provide a voice intercept announcement to the calling party when a call from Company's system is directed to a number that is unable to respond. is unable to respond. ΓC Γ 行机目目
 - Upon notification, Company will provide a reasonable period of inter-2. cept service when a call is directed to a number that has been absconnected from carrier's system. 86-84 1986 pile Service Commission

Effective: JUL

JUN 27 1988 Issued:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.1 REGULATIONS-(Continued)

32.1.19 Maintenance of Service Charges-(Continued)

- C. (Continued)
 - 4. After such date, a maintenance of facilities or service charge of \$25.00 will be assessed for each incident where trouble reported to carrier or Company as being on its system, facilities or equipment is later found to be on the system of, facilities or equipment of the reporting carrier or Company.
- 32.1.20 Obligations of the Carrier

Intercept and Acknowledgement of Calls by Carrier

- A. Outpulsing of Digits and End-to-End Signaling Arrangements
 - 1. Carrier shall provide a voice intercept announcement or distinctive tone signals to the calling party when a call is directed to a number that is not assigned by carrier.
 - Carrier shall provide a voice announcement or distinctive tone signals to the calling party when a call intended to activate a signal in a paging receiver has been received and accepted by carrier's terminal.
 - 3. When carrier's terminal is not able to complete calls because of a malfunction in the terminal or other equipment, carrier shall either divert the call to its operator or provide a recorded announcement to the calling party advising that the call cannot be completed.
 - 4. Carrier shall provide supervisory tones or voice announcements to the calling party on all calls, consistent with standard telephone industry practices. JUL 1 1900
- B. Line-Per-Terminal Systems
 - Interpersion and systems and a systems of the calling party when a call from Company's system is directed to a number that is unable to respond.
 - 2. Upon notification, Company will provide a reasonable period of 2155cept service when a call is directed to a number that has been disconnected from carrier's system.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

General Exchange Tariff -Section_32 Original Sheer FD DEC 2 9 1383 **EIISSOURI** Public Service Commission

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

General Exchange Tariff

ME UE

Section 32

DEC 20 1083

MISSOURI

Public Service Commission

JAN - 1 1984 83-253

Public Contro Com

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

> RADIO COMMON CARRIERS CANCELLED

32.1 REGULATIONS-(Continued)

32.1.20 Obligations of the Carrier-(Continued) DEC 1 1990

- BYLPYRS Sublic Service Commission B. Line-Per-Terminal Systems-(Continued)
 - When carrier's terminal is not able to complete calls because of 3. malfunction in the terminal or other equipment, carrier shall either divert the call to its operator or provide a recorded announcement to the calling party that the call cannot be completed.
 - 4. Carrier shall provide supervisory tones or voice announcement to the calling party on all calls, consistent with standard telephone industry practices.

32.2 SPECIAL CONSTRUCTION

This Section contains the regulations, liabilities, rates and charges 32.2.1 applicable for special construction of Telephone Company facilities which are used to provide services offered in this and other Southwestern Bell Telephone Company Tariffs.

> When special construction of facilities is required, the provisions of this Section apply in addition to all regulations, rates and charges set forth in other Southwestern Bell Telephone Company Tariffs.

32.2.2 Ownership of Facilities

DEC 29 1983

The Telephone Company retains ownership of all specially constructed facilities.

32.2.3 Interval to Provide Facilities

Based on available information and the type of service ordered, the Telephone Company will establish a completion date for the specially constructed facilities. The Telephone Company will make every reasonable effort to assure that the date is met. However, shortage of material, personnel or other factors may lengthen the installation interval. The Telephone Company does not guarantee that the facilities will be available on the scheduled date and assumes no liability if that date is missed. If the scheduled completion date=cannot be met,_ that date is missed. If the concerning date will be? established.

Issued:

JAN 0 1 1984 Effective:

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

32.2.4 Payments for Special Construction

A. Payment of Charges

MISSOURI All bills associated with special construction charges are due in accordance with the appropriate regulations in the service tariff under which service is being provided.

B. Start/End of Billing

Billing of recurring charges for specially constructed facilities starts on the day after the facilities are made available for use. Billing accrues through and includes the day that the specially constructed facilities are discontinued.

C. Partial Payments

To safeguard its interests during construction, the Telephone Company may require partial payment(s) for the portion of the estimated cost of the special construction for which a nonrecurring charge will apply. Partial payments may be requested as costs are incurred and will be credited against the total bill. Partial payments may not exceed the total nonrecurring charge for the special construction. If any partial payment is not received by the Telephone Company by the end of the first working day of the next month, the Telephone Company will cease all work on the special construction case, and cancellation charges will apply as set forth in Paragraph 32.2.5, D.7., following.

- 32.2.5 Liabilities and Charges for Special Construction
 - A. General

This Section describes the various charges and liabilities that may apply when the Telephone Company provides special construction of facilities in accordance with an order for service. Written approval of all liabilities and charges must be provided to the Telephone Company prior to the startof construction.

-253 Public Service Commissio

General Exchange Tariff

Or iginal She

CANCELLED

DEC.1 1990

Public Service Commission

Section 32

DEC 2 9 (203

MISSOURI

Public Service Commission

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

General Exchange Tariff CANCEL PEDnal Sheet USED RADIO COMMON CARRIERS DEC 1 1390 DEC 201003 BY CHASH Public Service Commission ISSOURI

FILED

32.2 SPECIAL CONSTRUCTION-(Continued)

32.2.5 Liabilities and Charges for Special Construction Continued ic Service Commission

B. Conditions Requiring Special Construction

Special construction is required when (1) suitable facilities are not available to meet an order for service, and (2) the Telephone Company constructs facilities and (3) one or more of the following conditions exist:

- The Telephone Company has no other requirement for the facilities constructed.
- It is requested that service be furnished using a type of facility, or via a route, other than that which the Telephone Company would normally utilize in furnishing the requested service.
- More facilities are requested than would normally be required to satisfy an order.
- It is requested that construction be expedited, resulting in added cost to the Telephone Company.
- C. Development of Liabilities and Charges

Special construction charges and liabilities will be developed based on estimated costs, except when actual costs are requested in writing prior to start of special construction. In order to meet a scheduled service date when actual costs are requested, an initial special construction case will be made based on estimated costs. Such case will be revised when actual costs are available.

D. Types of Liabilities and Charges

Depending on the specifics associated with each individual case, one or more of the following special construction charges and/or liabilities may be applicable:

1. Quotation Charge

A nonrecurring charge for the preparation of a quotation JAN = 1.1934applies whenever an estimate for special construction charge 3 - 253and liabilities is requested. The charge includes the costs 3 - 253associated with the development and preparation of the quotage commission tion.

Issued: **DEC 29 1983**

Effective: JAN 0 1 1984

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

32.2.5 Liabilities and Charges for Special Construction-(Continued): Service Commission

General Exchange Tariff

-Section-32 original sheet W2D

DEC 29 1983

I.11SSOURI

1390

DEC 1

83-253 Public Service Coramission

- D. Types of Liabilities and Charges-(Continued)
 - 1. Quotation Charge-(Continued)

The quotation is valid for 90 days and identifies all rates, charges and liabilities associated with the provision of the facilities required to satisfy a service request. Within this 90-day period, if the service is ordered as quoted and service is subsequently provided, the Ouotation Charge will be credited to the appropriate account.

If a request for a quotation is cancelled prior to its completion, a bill will be rendered for the costs incurred for quotation preparation through the cancellation date.

The payment of a Ouotation Charge does not assign, confer, or transfer title or ownership rights of proposals or facilities designed or furnished by the Telephone Company. Title and own- . ership rights for any item remains with the Telephone Company, except as specifically provided by a written agreement between all parties.

2. Nonrecurring Charge

A nonrecurring charge always applies and includes one or more of the following components:

Case Preparation Charge а.

> A nonrecurring charge always includes a case preparation associated with preparing a special construction case CANCELLED charge component to cover the administrative expenses

b. Expediting Charge

when it is requested that special construction be completed on an expedited basis. The charge equals the property of the second A nonrecurring charge may include an expediting charge expedited construction. JAN - 1 1924

Issued: DEC 29 1983 Effective: JAN 0 1 1984

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

32.2.5 Liabilities and Charges for Special Construction-(Continued)

- D. Types of Liabilities and Charges-(Continued)
 - 2. Nonrecurring Charge-(Continued)
 - c. Optional Payment

An optional payment charge may be included in the nonrecurring charge in association with a type of facility or route other than that which the Telephone Company would normally use in furnishing the requested service if lower recurring monthly charges are desired for the specially constructed facilities. This charge is equal to the excess installed cost or the total nonrecoverable cost, whichever is less. This election must be made in writing before special construction starts. If this election is coupled with the actual cost option, the optional payment charge will reflect the actual cost of the specially constructed facilities.

d. Replacement Charge

If any portion of specially constructed facilities for which an optional payment charge has been paid requires replacement involving capital investment, a replacement charge will apply. This charge will be in the same ratio to the total replacement cost as the initial optional payment charge was to the installed cost of the original specially constructed facilities. If any portion of the facilities subject to the replacement charge fails, service will not be restored until modification is provided in writing that replacement is required and such replacement is ordered.

e. Rearrangement Charge

If the Telephone Company is requested to rearrange explicitly constructed facilities, a nonrecurring \mathbb{ED} charge component equal to the cost of rearrangement will apply.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

LI Section 3220 Original Sheet 20 DEC 29 1983 [.1]SSOURI Public Service Commission

General Exchange [ariff]

CANCELLED

MISSOURI

83-253 Public Service Commission

1990

ice Commission

DEC 1

Public Serv

General Exchange Tariff

DEC 20 1000

MS_149

BEGiginal Sheet 21

Section .32

DEC 1 1990

Public Sérvice Commission

Public Service Counselssien

ShRS#1

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

- 32.2.5 Liabilities and Charges for Special Construction-(Continued). Min.Scien
 - D. Types of Liabilities and Charges-(Continued)
 - 2. Nonrecurring Charge-(Continued)
 - Special Construction of Facilities For Use For Less Than f. One Month

When the Telephone Company is requested to construct facilities to provide service for less than one month, a nonrecurring charge only applies. In addition to the case preparation charge component, this nonrecurring charge recovers all elements of cost, including engineering, shipping of equipment, equipment installation, lineup, equipment leasing, space rental, equipment removal, and any other costs associated with the construction of the facilities.

3. Maximum Termination Liability and Termination Charge

A Maximum Termination Liability is equal to the nonrecoverable costs associated with specially constructed facilities and is the maximum amount which could be applied as a Termination Charge if all specially constructed facilities were discontinued before the Maximum Termination Liability expires.

The liability period is equal to the average life of the account associated with the specially constructed facilities. The liability period is generally expressed in terms of an effective date and expiration date.

A Termination Charge may apply when all services using specially constructed facilities which have a tariffed Maximum Termination Liability are discontinued prior to the expiration of the liability period. The charge reflects the unamortized portion of the nonrecoverable costs at the time of termination, adjusted for net salvage and possible reuse. Administrative costs associated with the specific case of special construction and any cost for restoring a location to its original condition $J_{\rm AM}=1.193$? are also included. 83 - 253

Issued:

DEC 29 1983

JAN 0 1 1984 Effective:

General Exchange Tariff

RECEIVEN

DEC 2 Citos

Section 32

DEC 1 1990

JAN - 1 1934

83-253

Public

Original Sheet 22

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

- 32.2.5 Liabilities and Charges for Special Construction (Continued) SUURI
 - D. Types of Liabilities and Charges-(Continued)
 - 4. Annual Underutilization Liability and Underutilization Charge

Annual Underutilization Liability is a per-unit amount which is based on the per-unit cost of specially constructed facilities. The liability remains in effect until the expiration of the Maximum Termination Liability or until the special construction case is discontinued and all termination liabilities associated with the case are discharged. An underutilization charge may be applicable after the expiration of the minimum period, as set forth in the appropriate service tariff under which service is being provided, depending on the quantity of specially constructed facilities in service.

No underutilization charges are computed or billed until one year after the minimum period expires. At that time, an underutilization charge applies to the difference, if any, between the original number of specially constructed facilities and the number of specially constructed facilities in service at filed tariff rates. The underutilization charge applies from the date the minimum period expires and annually thereafter. For purposes of determining an underutilization charge, any facilities subject to minimum service period monthly charges are considered to be in service at filed tariff rates.

- 5. Recurring Monthly Charges
 - a. Excess Capacity Charge

A recurring monthly excess capacity charge applies when MISSOURI more facilities are requested and subsequently specially constructed than are required to satisfy an order for service. The charge is based on the estimated cost difference between the facilities constructed and the facilities which would normally be required to meet the order for service. Charges apply until there are sufficient services to warrant the facilities which were originally [[]]] []]

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

- 32.2.5 Liabilities and Charges for Special Construction-(Continued) CANCELLED
 - D. Types of Liabilities and Charges-(Continued)
 - 5. Recurring Monthly Charges-(Continued)
 - b. Charge for Route or Type Other Than Normal Public Service Commission

General Exhance fairff

Original Sheet 23

DEC 29 1983

ELISSOURI

Public Service Commission

DEC 1 1990 BY 104 R.S.#

83 - 253

When special construction is requested using a route or type of facility other than that which the Telephone Company would normally use, a recurring monthly charge, in addition to the monthly rates for service, is applicable. The charge is equal to the difference between the recurring costs of the specially constructed facilities and the recurring costs of the facilities the Telephone Company would have normally used.

- When an Optional Payment Charge as set forth in Paragraph 32.2.5, D.2.c., preceding, has been elected, the recurring monthly charge will be reduced to include specially constructed facility operating expenses only.
- (2) If the actual cost option as set forth in Paragraph 32.2.5, C., preceding, has been elected, the recurring charge will be adjusted to reflect the actual cost of the new construction when the costs have been determined. This adjusted recurring charge is applicable from the start of service.
- 6. Lease Charge

This charge applies when the Telephone Company leases equipment in order to meet service requirements. The amount of the charge is equal to the net added cost to the Telephone Company caused by the lease.

7. Cancellation Charge

If a service order with which special construction is associated is cancelled prior to the start of service, a Cancellation Charge will apply. The charge will include all nonrecoverable costs in Ecurred by the Telephone Company in association with the special construction up to and including the time of cancellation. JAN - 1 1934

Issued: DEC 29 1985

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.2 SPECIAL CONSTRUCTION-(Continued)

General Exchange Tariff Original Sheet 24 DEC 29 1983 MISSOURI Public Service Commission

JAN - 1 1984

32.2.6 Deferral of Start of Service

The Telephone Company may be requested to defer the start of service which will use specially constructed facilities subject to the provisions set forth in the service tariff under which service is being provided. Requests for special construction deferral must be in writing and are subject to the following regulations:

Construction Has Not Begun Α.

If the Telephone Company has not incurred any installation costs before receiving a request for deferral, the Quotation Charge applies. The original quotation is subject to Telephone Company review at the time of reinstatement to determine if the original charge estimates are still valid.

An additional quotation charge will, therefore, apply. Any change in liabilities and charges requires concurrence in writing.

B. Construction Has Begun

If the construction of facilities has begun before the Telephone DEC 1 1990 BY Lot RS follows: Public Service Commission

All Services Are Deferred 1.

When all services which will use specially constructed facilities are deferred, a charge based on the cost in the Company during each month of the deferral will apply. Those costs include the recurring costs for that portion of the facilities already completed and any other costs associated with the deferral. The Quotation Charge and cost of any components of the nonrecurring charge which have been completed at the time of deferral will also apply.

2. Some Services Are Deferred

> When some services which will use the specially constructed facilities are deferred, the construction case will be completed rand all 120 special construction charges will apply.

C. Construction Complete

If the construction of facilities has been completed before the tele 53 phone Company receives a request for deferral, all special Dionstruction mission charges will apply.

Issued: DEC 29 1900

Effective: JAN 0 1 1984

-

-

of cancel	ing this tariff.	as CANCE	LED
	RADIO COMMON CARRIER	-01	1990 20 1000
32.3 RAT		· C/V I	
32.3.1	Direct Inward Dialing Numbers	BY For the service MISS Monthly Rate	Nonrecurring Charge
	100 Direct Inward Dialing Numbers Assign <i>e</i> d (RDN)	\$13.00	None
32.3.2	Central Office Trunk Circuit Equipment		
	Central Office Trunk Circuit Equipment, per trunk (RTA)	Monthly Rate \$ 8.50	Nonrecurring Charge None
32.3.3		Monthly <u>Rate</u>	Nonrecurring Charge
	Intraexchange:		
	Local Channel (OPLAC)	\$26.50	\$30.00
	Interoffice Channel Terminal, each (OPLAT)	. 7.50	None
	Interoffice Mileage, per mile (OPLAM)	. 5.75	None
	Interexchange: Local Channel (OPLEC)	. 65.00	30.00
	Interexchange Channel Terminal, each (OPLET)		None
	Interoffice Channel Terminal, each (OPLEF)	. 7.50	None
	Interoffice Mileage, per mile (OPLEM)	. 5.75	None
	Interexchange Mileage, per mile (OPLEL)) 3.10	下作上民团
(See She	et 26 for Footnote.)		JAN - 1 1934 83 - 253 Public Sontice Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS

32.3 RATES-(Continued)

32.3.3 Outpulsing Connecting Circuit(1)-(Continued)

Footnote:

(1) The method of applying rates for interexchange and intraexchange service involving Radio Common Carriers (RCC's) are determined as follows:

Local Channel

A local channel charge shall apply for each channel terminated on the RCC's premises.

Interoffice Channel

Intraexchange - When station locations of an RCC service are located in different serving office areas, interoffice mileage charges apply. Charges are based on the V&H distance between the serving offices.

Interexchange - When a station location of an RCC service is located in an area served by a serving office that is not also the primary serving office, interoffice mileage charges apply. Charges are based on the V&H distance between the primary serving office and the serving office.

Channel Terminal

Channel Terminal Charges apply (1) at each termination of an interexchange channel and (2) each termination of an interoffice channel.

Interexchange Channel

When station locations of an RCC service are located in different exchanges or Metropolitan Exchanges, interexchange mileage charges apply.

Each exchange and each zone of a Metropolitan Exchange has one primary serving office. Charges are based on the V&H distance between primary serving offices.

CANCELLED	
DEC 1 1990	FIILED
BY ARS#1	'JAN - 1 1934
Public Service Commission MISSOURI	83-253 Public Service Commission

Issued: DEC 2 9 1983 Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

RADIO COMMON CARRIERS



32.3 RATES-(Continued)

32.3.4 Mobile Outdial Circuit

The carrier will be billed the applicable flat rate, one-party business line rates as provided in the Local Exchange Tariff, P.S.C. Mo.-No. 24.

32.3.5 Dial Access Line

The carrier will be billed the applicable flat rate or message rate, one-party business line rates as provided in the Local Exchange Tariff, P.S.C. Mo.-No. 24.

32.3.6 Line Per Terminal System

The carrier will be billed one-half the applicable flat rate, oneparty business line rate per line as provided in the Local Exchange Tariff, P.S.C. Mo.-No. 24.

32.3.7 Radio Transmitter Link, Special Signaling Circuits and Conditioning

The carrier will be billed the applicable circuit rates as provided in the Private Line Service Tariff, P.S.C. Mo.-No. 29.

32.3.8 Control Channels

The carrier will be billed the applicable circuit rates as provided in the Private Line Service Tariff, P.S.C. Mo.-No. 29.

32.3.9	Answer Supervision	Monthly Rate	Nonrecurring Charge
	Answer Supervision per trunk (ASSTT)(2)	\$15.25	\$140.00(1)

32.3.10 Miscellaneous Service Components

The carrier will be billed the applicable rates for miscellaneous service components not referenced in this Section of the Tariff, (i.e., Touch-Tone, Custom Calling Features, FX, 800 Service). These charges will be billed at the applicable rates according to the appropriate tariffs that cover each service component.

		CANCELLED	FILED I
(See Shee	et 28 for Footnotes.)	DEC 1 1990 BY ARS # 1 Public Service Commission	JAN - 1 1934 83 - 253 Public Service Pororeission
Issued:	DEC 2 9 1983	MISSOURI Effective: JAN 0 1 19	84
	-	RRON, Vice President-Missouri Tern Bell Telephone Company	

S't. Louis, Missouri

RADIO COMMON CARRIERS

32.3 RATES-(Continued)

Footnotes:

- This charge is controlled by each individual request from the carrier rather than the number of trunks per request.
- (2) Available in 1/1A and 2/2B ESS central offices only. The Bell Point of Contact must check with their switching engineering coordinate to determine if the trunk units are available, and if they are not available, whether the units are "hardwired" or plug-in. An SSAR should only be issued if a vendor must be scheduled into the central office to provide the proper trunk units. The trunk units will be added at no cost to the carrier if the carrier can wait until the vendor is scheduled into that office for our purposes (next-job basis). If the carrier does not wish to wait on a next-job basis, an SSAR should be established and processed. The carrier will incur charges only when the vendor must "hardwire" the trunk units to the frame. If the trunk unit is of the "plug-in" type, which can be installed in the central office without involvement of a vendor, no charge to the RCC will apply.

CANCELLED

DEC 1. 1990 st Public Service Commission MISSOURI



General Exchange Tariff

DEC 20 (083

MISSOURI

Public Service Commission

NEUE

Section 32

28

7 Sheet

Issued: DEC 29 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff		Gene	ral Exchange Tariff Section 32 Original Sheet 29
D.4.1			RECEIVED
KAL	DIO COMMON CARRI	ERS	
32.3 RATES-(Continued)			APR 25 1988
32.3.11 Digital Facility, 1.54	4 Mbps	Pul	MISSOURI blic Service Commission
	USOC	Monthly <u>Rate</u>	Nonrecurring Charge
<pre>- first 1/4 mile or fraction thereof, per facility</pre>	1LDP6	\$ 60.00	\$ 620.50
<pre>- each add'l 1/4 mile or fraction thereof, per facility</pre>	. 1LDPF	22.00	36.35
<pre>- flat rate usage per facility</pre>		366.00	
- digital termina- tion per facility.		77.50	

CANCELLED



FILED

JUN 8 1988 87-58 Public Service Commission

Issued:

APR 25 1988

Effective: MAI 2 6 1860

--

By R. D. Barron, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

General Exchange Tariff Section 32 1st Revised Sheet 30 Replacing Original Sheet 30

RADIO COMMON CARRIERS

RECEIVED

32.2 RATES-(Continued)

SEP 2 5 1989

32.3.11 Digital Facility, 1.544 Mbps-(Continued)

	USOC	Monthly Rate	Public Service (Nonrecurring <u>Charge</u>	and Equip- ment Charge
 4 Wire Interface (1) per equivalent voice grade channel 		\$32.80		
- E-M Signalling (1) - per equivalent voice grade channel		14.00	32.50	
- Answer Supervision (1) - per equivalent voice grade channel		15.25	140.00 (2)	
<pre>- Multiplexing (1) - per T-1 facility</pre>		409.55		(CR)5.50

32.3.12 Interoffice Truck Facility, 1.544 Mbps

	USOC	Monthly <u>Rate</u>	Nonrecurring Charge	Service and Equip- ment Charge
- fixed mileage per facility	1LNPN	\$180.50		
<pre>- per V-H mile or fraction thereof</pre>	1LNPX	5.75	CANCEL	LED

DEC 1 1990 BY Loh Public Service Commission

Effective:

(1) When digital facilities are used to terminate to an end office that is an analog switch, these charges apply in addition to the digital facility 1.544 Mbps charge and the flat rate usage charge. In this case, the digital termination charge of \$77.50 would not be applied.

2) Per individual request from the carrier, not per equivalent voice grade channel.

Issued: SEP 25 1985

COCT1 0 191989 89-14 By R. D. BARRON, President-Missouri Division rublic Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

General Exchange Tariff Section 32 Original Sheet 30

RADIO COMMON CARRIERS

32.3 RATES-(Continued)

RECEIVED

APR 2 5 1988

32.3.11	Digital Facility, 1.544 Mbps-(Contin	MISSOURI		
		Monthly	Public Service Commission	
	USOC	Rate	Charge	S&E
	- 4 Wire Interface (1) - per equivalent voice grade channel	\$ 32.80		
	- E-M Signalling (1) - per equivalent voice grade channel	14.00	32.50	. (
	- Answer Supervision (1) - per equivalent voice grade channet ANCELLE 1989	D 15.25	140.00 (2)	
	- Multiplexing (1) - per T-1 facility BY - Frice Co Public Service Co Public Service Co 1989 - MISSOL	· •		8.50
32.3.12	Interoffice Trunk Facility, 1.544 M	bps		
	USOC	Monthly <u>Rate</u>	Nonrecurring Charge	S&E

- fixed mileage per facility 1LNPN \$180.50 - per V-H mile or fraction 5.75 1LNPX thereof

- (1) When digital facilities are used to terminate to an end office that is an analog switch, these charges apply in addition to the digital facility 1.544 Mbps charge and the flat rate usage charge. In this case, the digital termination charge of \$77.50 would not be applied.
- (2) Per individual request from the carrier, not per equivalent voice grade channel. JUN 8 1988

APR 2 5 1988

Issued:

87-58 Public Service Commission Effective: MAY 25 1900

By R. D. Barron, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri